ResearchOnline@JCU

This file is part of the following reference:

Malekzadeh, Mona (2016) *Electrokinetic dewatering and consolidation of dredged marine sediments*. PhD thesis, James Cook University.

Access to this file is available from:

http://researchonline.jcu.edu.au/49416/

The author has certified to JCU that they have made a reasonable effort to gain permission and acknowledge the owner of any third party copyright material included in this document. If you believe that this is not the case, please contact <u>ResearchOnline@jcu.edu.au</u> and quote <u>http://researchonline.jcu.edu.au/49416/</u>



Electrokinetic Dewatering and Consolidation of Dredged Marine Sediments

Mona Malekzadeh

Submitted to the College of Science and Engineering in partial fulfilment of the requirements for the degree of

Doctor of Philosophy

James Cook University College of Science and Engineering November 2016

Statement of access

I, the undersigned, the author of this thesis, understand that James Cook University will make it available for use within the university library and, by microfilm or other means, allow access to users in approved libraries.

All users consulting in this thesis will have to sign the following statement:

In consulting this thesis, I agree not to copy or closely paraphrase it in whole or in part without the written consent of the author; and to make proper public written acknowledgement for any assistance which I have obtained from it.

Beyond this, I do wish to place any restriction on access to this thesis.

13/04/2017 Date and Signature

Statement of sources

DECLARATION

I declare that this thesis is my own work and has not been submitted in any form for another degree or diploma at any university or other institution of tertiary education. Information derived from the published or unpublished work of others has been acknowledged in the text and a list of references is given.

13/04/2017 Date and Signature

DECLARATION - ELECTRONIC COPY

I, the undersigned, the author of this work, declare that to the best of my knowledge, the electronic copy of this thesis submitted to the library at James Cook University is an accurate copy of the printed thesis submitted.

13/04/2017 Date and Signature

Acknowledgements

I would like to express my gratitude to all those who gave me the possibility to complete this research. I would like to express my special appreciation to my advisor Associate Professor Siva Sivakugan. I would like to thank you for all the encouragements and support throughout my research. I would also like to thank my co-supervisors Bobby Mathan, Peter To and Julie Lovisa for their help and contributions. I am also grateful for all the helps that Mr. Warren O'Donnell, Shaun Robinson has given me in the laboratory. At the end I want to thank undergraduate students Rahoi Buskens, James Wong, Daniel Robinson, and Benjamin Foster for their help in laboratory experiments.

Special thanks to my parent for their supports.

Abstract

Large amounts of marine sediments are produced every year as a result of capital and maintenance dredging works at major ports and these materials must be disposed of. Reuse of these dredged marine sediments for land reclamation is a sustainable and responsible method of disposal management. When the dredged sediments are utilized as a fill material in reclamation sites and are pumped into paddocks, there are two key concerns: 1) whether there is adequate storage capacity at the site 2) workability of the material. Since the dredged marine sediments possess a high water content, typically more than 250% and therefore are in the form of a slurry, self-weight settlement can take many years to complete. This is also due to the low permeability nature of the sediments. To improve the drainage of water and the settlement of these dredged muds, soil stabilization methods such as prefabricated vertical drains and vacuum preloading are commonly used. However, application of these methods is still time consuming and clogging during vacuum consolidation can be a limitation of this technique.

In this study the potential application of electrokinetic stabilization to improve the geotechnical properties of dredged mud in land reclamation areas is investigated as an alternative treatment method. In essence, electrokinetic stabilization is the application of electric potential to the soil through electrodes. When electric current occurs, the pore water flows from the anode to the cathode due to electroosmosis, soil particles move due to electrophoresis, and ions move toward the opposite charged electrode due to electromigration. The combination of all these three phenomena result in favourable and permanent changes in the soil. Electrokinetic stabilization is an environmentally friendly and time efficient method, however, a lack of standard design guidelines draws the attention of the geotechnical engineers away from this method.

In this study, key parameters influencing the efficiency of electrokinetic stabilization are investigated with the aim of maximizing the settlement of the dredged mud. For this purpose, a series of laboratory experiments have been undertaken to explore: (1) the most desirable electrode configuration, (2) effect of electrode materials, (3) the effective type and magnitude of the applied electric potential, and (4) the alteration of physiochemical properties. It is concluded that firstly, dredged marine sediments have significant potential to be improved by electrokinetic stabilization. Secondly, polyaniline coated anodes improve the technique at lower applied voltages. Thirdly, soil compressibility

reduces with depth, if the anode is placed at the bottom. Finally, higher settlement is achieved when the anode is placed at the bottom.

List of publications

Refereed Conference papers

Malekzadeh M., Lovisa J., Sivakugan N., Mathan B. (2014). "*Physiochemical changes during electrokinetic stabilisation of dredged mud*". 7th International Congress on Environmental Geotechnics (ICEG 2014). 10-14 November, Melbourne.

Malekzadeh M., Lovisa J., Sivakugan N. (2015). "*Consolidation of Cannington mine tailing*". The Changing Face of the Earth: Geomechanics & Human Influence- ANZ 2015 (ISSMGE). 22-25 February, Wellington.

Malekzadeh M., Sivakugan N. (2016). "Assessing the use of Aluminium electrodes on the electrokinetic dewatering and stabilization of dredged mud slurries". Sustainability, Energy, and Geo-environment (Geo-Chicago). 14-18 August, Chicago.

Journal Articles

Malekzadeh M., Lovisa J., Sivakugan N. (2016). "An overview of electrokinetic consolidation of soils". Journal of geotechnical and geological engineering. 34 (1), pp. 1-18, DOI: 10.1007/s10706-016-0002-1. Published.

Malekzadeh M. and Sivakugan N. (2016). "*Effect of one-dimensional electrode configuration on consolidation parameters of dredged mud*". Marine Geotechnology and Georesources. <u>Published.</u>

Malekzadeh M. and Sivakugan N. (2016). "Intermittent electro-consolidation of dredged sediments". International journal of geotechnical engineering. <u>Published.</u>

Malekzadeh M. and Sivakugan N. (2017). "Double drain electrokinetic stabilization of dredged mud using stainless steel electrodes and application of surcharge". KSCE Journal. <u>Published</u>

Malekzadeh M. and Sivakugan N. (2017). "Effect of polyaniline coated galvanized steel electrodes on electrokinetic sedimentation of dredged mud slurries". Canadian Geotechnical Journal. <u>Published</u>

Malekzadeh M. and Sivakugan N. "Effect of electrode placement on electrokinetic settlement". Environmental geotechnics. <u>Revised paper sent.</u>

Table of contents

Statement of accessi
Statement of sourcesii
Acknowledgementsiii
Table of contentsvii
List of tablesx
List of figuresxi
List of symbolsxii
1.1 Introduction
1.2 Aim and scope of the study2
1.3 Thesis overview
Chapter 2: Literature Review
2.1 Introduction
2.2 Electrokinetics
2.3 Background
2.4 Advantages and limitations of electrokinetic stabilization9
2.5 Electrokinetic stabilization design
2.6 Factors affecting electrokinetic applications in Soils
2.7 Methods to improve electrokinetic stabilization
2.5 General laboratory designed electrokinetic models
2.8 Theoretical framework for electrokinetic consolidation of soils
2.10 Summary
Chapter 3: Materials and preparations
3.1 General
3.1.1 Material: soil
3.2 Methods
3.2.1 Sample preparation
Conclusion
Chapter 4: Electrode configuration

4.1 Background	40
4.2 Horizontal electrode configuration	41
4.2.1 One-dimensional and two-dimensional electrokinetic stabilization	41
4.3 One-dimensional electrokinetic stabilization trial	43
4.3.2 Compression and rebound index	44
4.3.3 Coefficient of volume compressibility	45
4.4 Two-dimensional electrokinetic stabilization trial	46
4.5 Vertical electrode configuration, investigation on placement of the anode and	cathodes
4.5.2 Electroosmotic settlement	
4.5.3 Variation of electric current with time	53
4.5.4 Efficiency of the method	
4.5.5 Settlement versus energy consumption	
4.5.6 Dewatering versus energy consumption	55
4.5.7 Resistivity versus energy consumption	57
4.6 Effect of electrokinetic stabilization on consolidation parameters with depth	58
4.6.1 Compression index and Rebound index	60
4.6.2 Pre-consolidation pressure	61
4.7 Summary	62
Chapter 5: Type of electric current application	64
5.1 General	64
5.2 Effect of constant electric potential	65
5.2.1 Variation of electric current with time	66
5.2.2 Electrokinetic sedimentation	67
5.2.3 Power consumption	67
5.3 Effect of Irregular intermittent current application	68
5.3.2 Variation of electric current	68
5.3.3 Effect of Intermittent current application on soil sedimentation	70
5.3.4 Electroosmotic permeability	71
5.3.5 Effect of drainage and intermittent current on efficiency of the method	72
5.4 Summary	73
Chapter 6: Effect of electrode material	74
6.1 General	74
6.1.1 Conner	
VIII SUPPOR	

6.1.2 Aluminium mesh electrodes	76
6.1.3 Stainless steel electrodes	76
6.2 Galvanized steel and Polyaniline coated galvanized steel	77
6.2.1 Variation of current density with time	78
6.2.2 Electrokinetic sedimentation	81
6.2.3 Electrokinetic sedimentation coefficient	82
6.2.4 Electroosmotic permeability parameters	84
6.2.5 Power consumption	86
6.3 Summary	87
Chapter 7: Effect of electrode material on physiochemical properties	89
7.1 General	
7.2 Changes of pH	
7.3 Changes of electrical conductivity	91
7.4 Changes of salt content	92
7.5 Changes of liquid limit	93
7.6 Summary	94
Chapter 8: Summary, conclusion and recommendations	96
8.1 Summary	96
8.2 Conclusion	97
8.3 Further research	
References	

List of tables

Table 2.1 Collection of direct and coupled flow phenomena (After Mitchell 1993; Yeung and
Datla 1995; Horseman et al. 1996; Heister et al. 2004)7
Table 2.2 Practical effects of electrokinetic phenomena (Jones et al. 2006)7
Table 2.3 Historical development of electrokinetic treatment
Table 2.4 Review of the literature base on initial pH value13
Table 2.5 Electrical conductivity of the soils reported by researchers
Table 2.6 Boundary conditions and pore water pressure (Modified from Esrig 1968)25
Table 3.1 Mineralogy of the dredged mud obtained from port of Brisbane (modified from
Ganesalingam 2013)
Table 3.2 Metal elements in the dredged mud (Ganesalingam 2013)
Table 3.3 Physical properties of the dredged mud
Table 4.1 Compression index, rebound index, compression and recompression ratio of the treated
and untreated dredged mud45
Table 4.2 Compression index of the treated dredged mud with depth
Table 4.3 Preconsolidation pressure of the treated dredged mud with depth
Table 5.1 Power consumption versus applied voltage for application of constant voltages68
Table 5.2 Effect of intermittent current on soil resistivity and power consumption73
Table 6.1 Electrode materials used in this study to dewater and consolidate dredged marine
sediments74
Table 6.2 Optimum time at which maximum electric current occurs
Table 6.3 Power consumption, percent settlement and average electric current 87

List of figures

Figure 2.1 Advantages and limitations of the electrokinetic stabilization
Figure 2.2 Parameters influencing electrokinetic stabilization
Figure 2.3 Double layer on negatively charged clay particle
Figure 2.4 Methods to improve electrokinetic stabilization
Figure 2.5 Electrokinetic stabilization models presented in the literature
Figure 2.6 Electrokinetic stabilization models presented in the literature
Figure 2.7 Theoretical framework of electrokinetic stabilization
Figure 3.1 Plan view of port of Brisbane showing the location from which the samples were taken
Figure 3.2 Correlation between salinity of salinity obtained with probe and mass measurement a)
Port of Brisbane saltwater b) Strand saltwater
Figure 3.3 Correlation between salinity of the saltwater from port of Brisbane and the salinity of
the saltwater from Strand beach
Figure 3.4 Waterproof pH/Conductivity/Temperature tester probe with calibration kit
Figure 4.1 a) One-dimensional electrode configuration (2 array electrodes experimented in this
study) b) Two-dimensional electrode configuration - 4 cathodes, one anode in the centre is
experimented herein (Alshawabkeh et al. 1999b)42
Figure 4.2 a) The electric field between two non-identical electrodes b) The electric field s
between two-identical electrodes (Ogborn et al. 2000)
Figure 4.3 Schematic view of one-dimensional electrokinetic stabilization set-up
Figure 4.4 Void ratio versus effective stress of the stabilized dredged mud45
Figure 4.5 Coefficient of volume compressibility versus effective stress
Figure 4.6 Plan view of two-dimensional electrokinetic stabilization set-up47
Figure 4.7 The vane shear apparatus
Figure 4.8 Shear strength profile after electrokinetic stabilization
Figure 4.9 Water content profiles after electrokinetic stabilization
Figure 4.10 The relationship between water content and undrained shear strength
Figure 4.11 Settling profile in one-dimensional and two-dimensional profile and generation of
desiccation cracks
Figure 4.12 Vertical electrokinetic stabilization set up based on settlement column
Figure 4.13 Changes of interface height versus log time
Figure 4.14 Variation of electric current with time
Figure 4.15 Electroosmotic and electrophoretic settlement of the dredged marine slurry

Figure 4.16 Amount of dewatering with time during electrophoresis and electroosmotic
settlement
Figure 4.17 Efficiency of electroosmosis
Figure 4.18 Method of taking samples from bottom of the column
Figure 4.19 Location of the sampling with depth for oedometer testing
Figure 4.20 Compression curves of the treated dredged mud with depth
Figure 5.1 Variation of electric current with time for different voltages
Figure 5.2 Electrokinetic surface settlement of the dredged mud with application of different
electric potential
Figure 5.3 Set-up used to investigate the effect of intermittent current
Figure 5.4 Variation of electric current with time for double drained intermittent current (DDIC),
single drained intermittent current (SDIC), and single drained constant voltage (SDCV) 70
Figure 5.5 Settlement versus time71
Figure 5.6 Electroosmotic conductivity versus void ratio relationships72
Figure 6.1 The copper anode corroded such that it consumed and disintegrated completely 75
Figure 6.2 Corrosion of the aluminium anode (a) and aluminium cathode (b) after electrokinetic
stabilization76
Figure 6.3 Variation of electric current with time77
Figure 6.4 Circular galvanized steel
Figure 6.5 Variation of electric current with time during electrokinetic stabilization of dredged
mud slurry79
Figure 6.6 Variation of electric current with settling velocity during electrokinetic stabilization
of dredged mud slurry80
Figure 6.7 Changes of settlement versus time
Figure 6.8 Variation of settling velocity with porosity
Figure 6.9 Electroosmotic permeability versus void ratio
Figure 7.1 a) Variation of pH with depth b) Zone of consolidation91
Figure 7.2 Variation of electrical conductivity with depth
Figure 7.3 Variation of salt content with depth
Figure 7.4 Changes of liquid limit with depth

List of symbols

A: are of the specimen

- Cc: compression index
- CR: compression ratio
- Cr: recompression index

DDIC: double drain intermittent current

DM-AL: dredged mud aluminium

DM-CA: dredged mud coated anode

DM-GS: dredged mud galvanized steel

E: electric field intensity

e: Void ratio

e: void ratio

EC25: electric conductivity on temperature factor

ECt: electric conductivity on temperature factor

EO: electroosmosis

ft: temperature factor

G_s: Specific gravity

I: Amps (electric current)

I: hydraulic gradient

j: current density

k: Permeability

ke: electroosmotic permeability

k_h: hydraulic conductivity

L: distance between electrodes

LL: Liquid limit

P: Power consumption per unit volume

PL: Plastic limit

Qe: rate of water flow due to electric current application

Q_h: water flow rate

r: sedimentation coefficient

RR: recompression ratio

SDCV: single drain constant voltage

SDIC: single drain intermittent current

Su: undrained shear strength

U: pore pressure

- V: Volts (electric potential)
- Vs: Settling velocity
- W: energy consumption
- γ_s : Unit weight of solids
- γ_w : Unit weight of water
- π : Intermittent ratio
- ρ_s : Density of solid
- ρ_w : Density of water
- σ ': effective stress
- Ω : electrical resistivity

Chapter 1: Introduction

1.1 Introduction

Land reclamations are often created as a storage place for beneficial use of dredged marine sediments. Dredged marine sediments are ultra-soft soils possessing high fine content and high moisture content with low permeability due to which their self-weight settlement takes a long time to complete. Different methods of soil improvement such as chemical stabilization, prefabricated vertical drains, and vacuum preloading have been used to accelerate the consolidation settlement of the soil. Chemical stabilization using lime, fly ash, slag, and cement admixtures are used to improve the engineering properties of soft soils (Mitchell 1976, Supakij et al. 2004, Ho and Chan 2011).

To add admixtures to the soil, factors such as curing time, temperature, soil type, initial moisture content, mixing method to obtain a homogenous soil need to be considered. Overall, addition of the chemicals to the soil may increase the strength of the soil, however, this does not contribute to the higher settlement of the material as a result of which more storage capacity can be obtained. Therefore, chemical stabilization does not contribute to the increase in rate of settlement. Prefabricated vertical drains increase the rate of settlement by improving the drainage of the soil. However, the prefabricated vertical drains undergo a high deformation due to the high compressibility of soft soils. Vacuum preloading is another method for accelerating the settlement and drainage of soft soils with high water content. The vacuum preloading increases the effective stress of the soil by applying suction. This suction pressure drains and imparts a hydraulic gradient that prevents the generation of excess positive pore-water pressure (Indraratna et al. 2005). However, clogging of the drains during the vacuum preloading is not desirable.

Electrokinetic stabilization is an innovative technique where the electric current is applied to the soil through electrodes where pore water moves from the positive electrode (anode) to the negative electrode (cathode). This method is used for slope stability, excavations, increasing the capacity of piles, mitigating liquefaction potential of silty soils, dewatering sludge (Soderman and Milligan 1961; Chappell and Burton 1975; Burnotte et al. 2004; El Nagger and Routledge 2004; Chen and Jia 2007). However, the

use of electrokinetic stabilization for improving the properties of the dredged mud has not been investigated.

The use of electrokinetic stabilization for geotechnical purposes is pioneered by Casagrande (1961). However, the lack of standard design and unknown changes in chemical properties of soils draws the attention of geotechnical engineers away from this method. The accurate analysis of electrokinetic stabilization requires understanding of the process and its effect on changing the properties of the soil.

1.2 Aim and scope of the study

The primary objective of this study is to firstly investigate the effect of electrokinetic stabilization on properties of the dredged mud and to investigate its potential application and the response of dredged mud to the method. Secondly, it is proposed to investigate the upper limit and maximize the efficiency and reduce the cost of the method through understanding the effect of type of electric current application, electrode configuration, electrode type and material and its effect.

The sub-objective of this study comprised of the following:

- Optimizing the efficiency of electrokinetic stabilization of dredged marine sediments based on the response of soil to the method.
- Mitigating the corrosion rate of the metal anode through application of coating materials.
- Investigating the ideal location and distance between electrodes which is unclear in the literature.
- Investigating the effect of the type of electric current application and its effect on efficiency of the electrokinetic stabilization method.
- Studying the effect of different electrode configurations including onedimensional and two-dimensional electrokinetic stabilization.
- Investigating the effect of electrode material.

1.3 Thesis overview

Chapter 1 explains the limitations in the beneficial use of dredged marine sediments for land reclamation purposes. The current practices and the methods to reduce the timeframes of the soil sedimentation is given. A brief overview of this thesis is also presented.

Chapter 2 emphasizes the recent developments in electrokinetic stabilization and factors affecting the process of electrokinetic stabilization, methods to improve the efficiency of the method, and current electroosmotic consolidation models and their boundary conditions.

Chapter 3 presents information about physical properties of the dredge mud, its mineralogy, and the method of sample preparation for electrokinetic stabilization testings along with the procedure for measuring the physiochemical properties of the dredged mud before and after treatment.

Chapter 4 focuses on how electrode configuration affects the electrokinetic stabilization. Horizontal configuration (one-dimensional and two-dimensional), and vertical configuration (cathode top-anode bottom; cathode bottom-anode top) are investigated. The investigation is carried out through understanding the effect of electrokinetic stabilization on settlement of the soil. The changes of consolidation parameters with depth is also presented in this chapter.

Chapter 5 describes how type of electric current application changes the electrokinetic stabilization of soils. Effect of irregular intermittent current application and the drainage condition is presented in this chapter.

Chapter 6 investigates the effect of electrode material. Different type of materials copper, stainless steel, galvanized steel, and polyaniline galvanized steel are tested. To reduce the corrosion rate of the anode polyaniline coating is applied to the anodes, and the changes in efficiency of the electrokinetic stabilization is shown.

Chapter 7 shows the change of physiochemical properties of the soil such as pH, electric conductivity, soil salinity, and liquid limit after electrokinetic stabilization for and their interpretation.

Chapter 8 presents conclusions of this study along with further recommendations for future research in this area.

Chapter 2: Literature Review

2.1 Introduction

The need for soil stabilization is rising due to an increase in global population and limitation of appropriate lands for construction purposes. Geotechnical properties of soils provide an indication of their suitability for construction purposes. Soils with poor geotechnical properties (low compressive and shear strengths, low bearing capacity, etc.) require stabilization. Failure to do so results in extensive damages to future structures, which would be so costly to repair or to mitigate the impacts.

Limitation of appropriate lands coupled with the need for disposal sites for industrial wastes necessitate the development of methods to improve soils. For beneficial use of slurries such as dredged mud and mine tailings, the first approach is to reduce its water content until it gains enough strength of 1 kPa, to be considered as a soil (Azam et al. 2007). Then soil stabilization is an option to improve the soil furthermore. For this purpose, a range of different methods of soil stabilization have been developed for soft soil conditions such as:

- soil densification,
- physical and chemical stabilization,
- modification with inclusions, and
- hydraulic modification.

Soil densification through compaction is one of the most common method of soil improvement in field. Densification through compaction can be shallow or deep depending on the method and the equipment. Shallow compaction affects the top surface layer of the soil and deep compaction affects the soil in deeper levels. This method of soil modification cannot be applied to a soft soil with high moisture content, which is in the form of a slurry.

Chemical stabilization is through in-situ deep mixing of chemicals such as cement, lime, fly ash, or steel slag with the soil for strengthening purposes. However, with this method, curing time is an important factor to consider which in turn renders it rather timeconsuming to achieve the appropriate strength. The homogenous mixing of the chemicals in practice is another drawback of this technique.

Modification with inclusions is using geosynthetic fibres for soil reinforcement. This method improves the strength properties of soils and reduces the shrinkage properties in case of expansive soils. Geosynthetic fibre is used in combination with chemical stabilization to increase its effectiveness. However, it is difficult to distribute these fibres evenly in practice.

Hydraulic modification of the soil is through changing the flow of water. This involves drainage, dewatering, and seepage or groundwater flow. Slurries such as dredged mud and mine tailings have very high water content. Settlement of these slurries is time consuming and can be accelerated using a number of techniques, most notably, by hydraulic modifications such as prefabricated vertical drains or vacuum preloading. Any of these techniques consolidates soils either by removing water or air from the system. Application of surcharge is often done simultaneously with the application of vertical drains to improve the efficiency of the technique. Vacuum preloading consolidates soil by removing air and water. Although, it is a faster method than the conventional vertical preloading, failure due to insufficient undrained bearing capacity is a common concern of this technique. An innovative method of soil stabilization known as electrokinetic stabilization with which, consolidation is facilitated by application of direct electric current to the soil.

Electrokinetic stabilization improves the geotechnical properties of the soils including shear strength, compressibility, and permeability. However, due to the lack of understanding and proper design criteria, this method requires extensive research before becoming viable as another method of soil stabilization for applications such as improvement of dredged marine sediments and if shown successful, to determine an efficient design model.

This thesis is focused on an investigation of the suitability of electrokinetic stabilization for improving the settlement and geotechnical properties of the dredged marine sediments. In this chapter, an overview of literature and current developments including the background, design parameters, and factors influencing the electrokinetic stabilization presented.

2.2 Electrokinetics

Electrokinetic stabilization is an application of electrical current to the soil through the electrodes. Electrokinetic stabilization includes a combination of following processes:

- electro-osmosis,
- electrophoresis,
- electrolysis, and
- electro-migration

which collectively change the structure of the soil.

Electro-osmosis is the transport of water from anode to the cathode. This is due to the influence of electric field on dipole water molecules that results in movement of capillary water. Electrophoresis is the movement of charged particles. Electrolysis is the decomposition of the pore water at electrodes. The electrolysis of water occurs due to the redox reactions (oxidation-reduction reactions), given as follows:

$$2H_2O - 4e^- = 4H^+ + O_2 (Anode)$$
(2.1)

$$2H_2O + 2e^- = 2OH^- + H_2 (Cathode)$$
(2.2)

These reactions result in generation of oxygen and hydrogen gas around the anode and the cathode, respectively (Table 2.2). The oxidation reaction results in reduction of pH and soil plasticity near the anode. Reduction reaction increases the soil pH near the cathode due to the dissolution of hydrogen ions.

Electro-migration is movement of ions due to the induced electric potential, resulting in movement of solute such as salt during the electrokinetic process. These electrokinetic phenomena affect the soil. The practical outcomes of the electrokinetic components are presented in Table 2.1.

Flow	Equation	Gradient X			
		Hydraulic	Temperature	Electrical	Chemical
Fluid (q _h)	$q_h = k_h \frac{\Delta H}{L} A$	Hydraulic conduction (<u>Darcy's law</u>)	Thermo-osmosis	Electroosmosis	Chemical- osmosis
Heat (q_t)	$q_t = k_t \frac{\Delta T}{L} A$	Isothermal heat transfer	Thermal conduction (<u>Fourier's law</u>)	Peltier effect	Dufour effect
Current (<i>l</i>)	$I = \sigma_e \frac{\Delta V}{L} A$	Streaming potential	Thermoelectricity (Seebeck effect)	Electric conduction (<u>ohm's law</u>)	Diffusion and membrane potentials
Ion (J _D)	$J_D = nD \frac{\Delta c}{L} A$	Streaming potential	Thermal diffusion of electrolyte (Soret effect)	Electrophoresis	Diffusion (<u>Fick's</u> <u>law</u>)
where $\frac{\Delta H}{L}$, $\frac{\Delta T}{L}$, $\frac{\Delta V}{L}$, $\frac{\Delta c}{L}$ are hydraulic gradient, thermal gradient, electric potential gradient, and ion flux due to which the mentioned flow occurs. k_h , k_t , σ_e , nD are hydraulic conductivity, thermal conductivity, electrical conductivity, and diffusion. A is the cross-sectional area of the sample.					

Table 2.1 Collection of direct and coupled flow phenomena (After Mitchell 1993; Yeung andDatla 1995; Horseman et al. 1996; Heister et al. 2004)

During electrokinetic stabilization water, electric current, heat and ions flow through the soil. These flows are due to hydraulic, temperature, electric or chemical gradient that are the result of electrokinetic phenomena. During electrokinetic stabilization, these flow in the soil couple and change the properties of the soil (Table 2.2).

Table 2.2 Practical effects of electrokinetic phenomena (Jones et al. 2006)

Electrokinetic phenomenon	Effects
Electroosmosis	Water flow, Pore pressure change, Volume change
Electrophoresis	Particle movement, particle sedimentation
Electromigration	movement of dissolved substance (like salt) – migration of ions
Electrolysis of water	Oxygen evolution, Hydrogen evolution, pH changes

2.3 Background

Electro-osmosis was first reported by Reuss (1809). He was the first to observe the flow of water from the anode to the cathode by application of electrical current to the saturated clay. Later, Quincke (1861) improved the understanding by describing the flow potential. After Reuss' work, Perrin (1904) and Smoluchowski (1921) established Helmholtz-Smoluchowski (H-S) equation, which shows the electroosmotic parameters and their interrelationships.

The first successful application of electroosmosis in geotechnical applications was trialled by Casagrande (1949). The studies in the literature start mainly from electroosmosis which is the flow of pore water from anode to the cathode. To enhance the application of electrokinetic stabilization on different soil types, this method has been studied intensively during the last 80 years. However, due to the lack of a suitable standard design approach, the application of electrokinetic stabilization requires a laboratory simulation. The historical development of the electrokinetic applications is given in Table 2.3.

Year	Researcher	Observations and studies
1809	Reuss	First researcher that observed electroosmosis by application of direct current to mixture of clay and water
1846	Napier	Recognized the distinction between electroosmosis and electrophoresis
1861	Quincke	Studies the flow and streaming potential
1879	Helmholtz	Provided analytical model for electroosmosis
1904	Perrin	Derivation of electrokinetic velocity
1921	Smoluchowski	
1949	Casagrande	Studied electrokinetic phenomena in porous media like soils and performed the first practical studies
1960	Russian researchers	Electromigration of metals
1961	Soderman and Milligan	Increasing bearing capacity of piles with electrokinetic stabilization
1963	Veder	Electrokinetics of bentonite slurry wall
1975	Chappel and Burton	Stabilization of slopes and excavations
1980	Banerjee and Mitchell	Determination of electroosmotic coefficient of consolidation
1986	Dutch researchers	Successful application of electrokinetic stabilization (soil remediation)

 Table 2.3 Historical development of electrokinetic treatment

1980 - 1998	Researchers around	Successful application of electrokinetic treatment	
	the world	for decontamination purposes	
1996	Shang and Dunlap	Using electrokinetic stabilization to increase soil	
		strength	
1997	Shang	Reversing the polarity to increase the homogeneity	
		of the soil after electrokinetic stabilization	
1998	Shang	Developed 2D electroosmotic consolidation	
2001	Micic et al.	Application of intermittent current to improve the	
		electrokinetic stabilization	
2004	Pugh and Jones	Patent for innovative electrodes known as EKGs	
	-	(electrokinetic geosynthetics)	
2005	Asavadorndeja and	Anode depolarization	
	Glawe		
2005- Now	Yang et al. (2005),	Investigating and analyzing the effect of	
	Kaniraj and et al.	electrokinetic stabilization on different type of	
	(2011), Fourie and	soils and developing methods to reduce the	
	Jones (Reuss,	limitation of the method such as power	
	1809)(2010), Lee	consumption.	
	et al. (2013), and		
	other researchers.		

2.4 Advantages and limitations of electrokinetic stabilization

Electrokinetic stabilization is an environmentally friendly and promising method to dewater and consolidate slurries and soils with low permeability. A brief explanation on the advantages and limitations of the electrokinetic stabilization method is given in Figure 2.1. Especially in the case of underwater structures where application of other methods is not possible due to the high water content of the soil, electrokinetic stabilization has potential application. However, due to the lack of design criteria, there is a need for further research to maximize the efficiency and effectiveness of the method.

2.5 Electrokinetic stabilization design

Despite the extensive research on electrokinetic stabilization, there is a lack of design criteria in the literature. The lack of suitable design criteria is due to the factors influencing the electrokinetic stabilization method. These factors change during electrokinetic stabilization and result in changes of soil structure. The energy consumption of the method changes depending on the changes of electrokinetic parameters such as physiochemical changes in the soil and electrical parameters. The effect of these parameters on electrokinetic stabilization is discussed in next section.

Advantage	 It can be applied without disturbance to the adjacent areas, and can be applied to the constructed lands to enhance the soil beneath foundations It can treat wide range of soils especially very effective for treatment of clayey soils, due to their negative surface charge and low hydraulic conductivity. It has multiple objectives of consolidation, strengthening and decontamination.
Limitatio —	 A reduction in electrical current density which is due to the activation, resistance and concentration polarization (Acar and Alshawabkeh 1993, Alshawabkeh and Acar 1996; Acar et al. 1995; Sah and Chen 1998). Generation of bubbles of O₂ and H₂ around electrodes, which are the result of oxidation and reduction reactions near anode and cathode, result in the loss of soil-electrode contact (Bergado et al. 2003). An increase of soil resistivity to the same applied voltage and the need to increase the voltage that leads to a higher power consumption (Mohamedelhassan and Shang 2001). Unknown electrochemical reactions. Lack of standard design.

Figure 2.1 Advantages and limitations of the electrokinetic stabilization

2.6 Factors affecting electrokinetic applications in Soils

The physio-chemical properties of soil change due to the electrochemical reaction during electrokinetic stabilization. Factors influencing the electrokinetic process are divided into two main groups. First group is related to the initial soil condition and second group is set-up design parameters. Figure 2.2 shows a brief explanation on parameters influencing the electrokinetic stabilization.

Water content

The initial water content affects the soil porosity, void ratio and electrical resistivity of the soil. Fourie et al. (2007) reported 83% water content reduction after application of electroosmosis dewatering of mine tailings with initial gravimetric water content of 257%. Water content in peat and clayey soils with an average initial value of 472% and 355% was reduced significantly using electroosmotic consolidation (Kaniraj et al. 2011). This is a significant improvement. Depending on the properties and type of soil,

successful application of electrokinetic stabilization with different initial water contents is reported in the literature (e.g. Kaniraj et al. 2011; Liaki et al. 2010; Chien et al. 2011; Jayasekera and Hall 2007). In general, 1.6 times of the soil liquid limit was found as the most common ratio (Jeyakanthan et al. 2011).

During electroosmosis, as the water content of the soil reduces, desiccation cracks appear. These cracks reduce the efficiency of the electroosmosis. It is necessary to know at which value of water content soil starts to crack. For dredged mud the water content at which it starts to shrink during self-weight sedimentation, which is known as desiccation limit, is estimated at 1.2 times its plastic limit (Stark et al. 2005). This water content should be taken into consideration to determine the initial moisture content that is appropriate for electrokinetic stabilization. Therefore, dredged mud with water content less than desiccation limit is inappropriate for electrokinetic stabilization even for purpose of soil strengthening.

pH and zeta potential

The next factor is pH, which is defined as the ability of the soil to react with chemical admixtures, because some reactions can occur only at a specific pH value. Soil's ability to withstand pH fluctuations is expressed as soil buffering capacity. Soils with high buffering capacity often have high clay and organic content (Burton 2010). However, the alteration of soil buffering capacity, as well as pH, can also be done by adding base or acid. Soils with high buffer capacity are preferable for electrokinetic stabilization, since they are able to withstand the pH fluctuations, which normally occur during the process. Table 2.4 illustrates the initial pH value of different soils that electrokinetic stabilization is performed on.

The acidity or alkalinity of the soil is based on the stability of the ions at different pH values. For instance, metal cations are stable under acidic environment, while anions are stable under alkaline condition (Hamed et al. 1991). Thus, the control of pH is important since it can affect the movement of ions. During electrokinetic stabilization, an acidic environment (high pH value) near anode is generated due to the existence of H⁺ ion and oxygen gas and an alkaline environment is generated near cathode due to the existence of OH⁻ ions and hydrogen as a result of electrolysis of water.



Figure 2.2 Parameters influencing electrokinetic stabilization

Effect of constant voltage is experimented by Lockhart and Hart (1988). Based on his study the overall dewatering efficiency is less when higher voltages are applied (Lockhart 1993). Potential gradient more than 1 V/cm is not efficient.

Changes in pH affect the particle stability, solubility, dispersion, precipitation and chemical reactions in soil. The tendency of hydroxyls to dissolve in water is strongly affected by pH. When pH of the clay increases or reduces sharply, soil compositions such as iron, aluminium, sodium disperse and later precipitate as hydroxides or salt. Liaki et al. (2010) showed that if pH of kaolinite reduces below 5, the aluminium ions migrate and cause soil strengthening. On the other hand, the formation of stabilizing agent occurs under alkaline conditions. Pozzolanic reaction occurs at alkalinity and the cementing agent results in an increase of soil strength and changes of Atterberg limits. Jayasekera and Hall (2007) observed an increase in liquid limit and plastic limit near the cathode (alkaline environment) and reduction near the anode (acidic environment).

Zeta potential is the electric potential in a double diffuse layer. Diffuse Double layer (DDL) is a charged layer around the wet clay particle (Figure 2.3). Since pH and zeta potential have a direct relationship, the changes in pH cause changes in zeta potential and electrical conductivity. Acar et al. (1989) and Hamed et al. (1991) observed that the reduction of pH results in reduction of zeta potential and hydraulic conductivity, which in turn slows down the flow of fluid through the soil mass. In electrokinetic stabilisation, pH changes in each part of the soil mass depends on the conductivity of the presented ions. Sharp changes of pH and existence of insoluble hydroxyls during electrokinetic stabilization cause a drop in the electric potential in the soil (e.g Hamed et al. 1991; Liaki et al. 2010).

Author	Type of the soil	Initial pH
Shang (1996)	Brown clay, Grey clay	7.4, 7.2
Hamir et al. (2001)	Kaolin grade E	5.0 ± 0.5
Reddy et al. (2006)	Dredged sediment	7.0
Fourie et al. (2007)	Sand tailing	6.4
Jayasekera and Hall (2007)	Silt loam – heavy clay	4.5
Rittirong et al. (2008)	London Ontario clay	7.5
Kamarudin et al. (2008)	Residual soil	5.3
Kalumba et al. (2009)	Soft clay	8.2
Loch et al. (2010)	Na-Bentonite clay	7.5
Kim et al. (2011)	Dredged sediment	7.7

Table 2.4 Review of the literature base on initial pH value



Figure 2.3 Double layer on negatively charged clay particle

Level of soil salinity

Soil salinity affects the electroosmotic flow by affecting its zeta potential (Mitchel and Soga 2005). Soils with higher zeta potential have higher electroosmotic permeability. When soil salinity increases, the zeta potential reduces due to the reduction of double layer thickness. Therefore, electrokinetic stabilization is unlikely to be successful in soils with very high salinity. Hence, it is important to determine the salinity of the soil prior to the electrokinetic application.

The level of soil salinity is estimated by either electrical conductivity (S or mS), direct measurement of salt content (ppm), or measurement of the mass of the salt crystals. There are many controversies on the specific limit for soil salinity that result in optimum electrokinetic stabilization. According to Bergado et al. (2000) soils with salinity higher than 6000 ppm are not responsive to electrokinetic stabilization. Mitchel (1991) suggests that soils with electrical conductivity above 0.003 S/cm are responsive to electrokinetic stabilization. Jones and Glendinning (2006) reported that electroosmotic stabilization can be effective and economical if soil electrical conductivity is between 0.050 S/cm – 50 S/cm. However, depending on the type of the soil, lower initial electrical conductivity might also be applicable if methods such as addition of chemical admixtures and salt solution are used. Micic et al. (2001) suggest that

soil with salinity of 0.003 S/cm or less, result in successful application of electrokinetic stabilization. On the contrary, Mohammedelhassan and Shang (2002) observed successful application of electrokinetic stabilization of soils with 0.013 S/cm Nacl. Micic et al. (2001) has experimented with electrokinetic application on marine sediments with 0.050 S/cm of Nacl.

Electrical resistivity and conductivity

Soil electrical conductivity σ (measured in milli-Siemens/cm or Siemens/m) is the ability of the soil to transfer electrical current. The unit Siemen is defined as Ω^{-1} or Ampere per volt. Soil resistivity, which is the soil's ability to resist electrical current, has an inverse relationship with soil electrical conductivity. Soil resistivity is measured in Ω ·m or Ω ·cm. Electrical resistivity of soils can be affected by many factors such as temperature, water content, soil porosity, pore fluid resistivity, the soil composition, soil salinity, particle and pore size and shape (Yan et al. 2012). Soils with very low water content are not able to conduct enough electrical current. A critical water content above which soil demonstrates electrical conductivity is 15% (Sverko, 1999).

Electrical resistivity is varied by soils due to the existing minerals, chemical and electrical properties such as dielectric constant and zeta potential. For instance, surface soils and clay soils have low resistivity and high conductivity; this might be attributed to the existence of carbon in surface soils and negative surface charges in clays. Electrical conductivity of the soil can be expressed in terms of the soil electrical resistivity by the following equation or it can also be estimated from the plasticity index of the soil.

$$\sigma = \frac{1}{R} \frac{L}{A} \tag{2.3}$$

where σ (Siemens) is the electrical conductivity, $R(\Omega)$ is electrical resistivity, L(m) is length and $A(m^2)$ is the cross sectional area of the sample.

During electrokinetic stabilization, the ionic concentration changes due to electromigration. Therefore, electrical conductivity of the soil varies based on the ionic strength. As a result, different zones within the soil may have different values of electrical conductivity. This change in local electric conductivity results in variations in electric potential. The initial electric conductivity shows the initial ionic concentration in the soil. Therefore, it is important to know the electrical conductivity of the soil prior to electrokinetic stabilization. Table 2.5 shows different values of initial electrical conductivity that are reported in the literature.

		Electrical
Author	Soil Type	conductivity (S/cm)
Shang (1997)	Grey clay	0.000289
Hamir et al. (2001)	Kaolin	0.025
Micic et al. (2001)	Yulchon clay	0.012
Glendinning et al. (2005)	Clay	0.006
Jayasekera and Hall (2007)	Silt loam	0.002
Jayasekera and Hall (2007)	Heavy clay	0.003
Kalumba et al. (2009)	Tunnel Slurry	0.23
Kim et al. (2011)	Sediments	0.000842
Fourie et al. (2007)	Sand tailing	0.00051
Rittirong et al. (2008)	London Ontario clay	0.000222
Loch et al. (2010)	Clay	0.001

Table 2.5 Electrical conductivity of the soils reported by researchers

Electrode material

The material constituting the electrodes affect the electrokinetic stabilization. Different types of electrode material such as inert metals, non-inert metals, and carbon based electrodes are used for electrokinetic stabilization of soils (e.g. Liaki et al. 2010; Abdullah and Al-Abadi 2010; Kaniraj et al. 2011). When a metal electrode is used, the anode corrodes due to electrolysis. The electrolysis and corrosion of the anode occur due to the following reactions:

$$M + H_2O = MO + 2H^+ + 2e^-$$
(2.4)

where M is the metal element and MO is metal oxide.

For instance, if iron (Fe), copper (Cu) or carbon (C) electrodes are used, the following reactions occur near anode which cause the corrosion of electrodes and result in different chemical reactions within the soil.

$$Fe + H_2O = FeO + 2H^+ + 2e^-$$
 (2.5)

$$Cu + H_2O = CuO + 2H^+ + 2e^-$$
(2.6)

$$C + H_2O = CO_2 + 4H^+ + 4e^-$$
(2.7)

This can be added to the oxidation reaction. The following equation is used to investigate the amount of corrosion at the anode.

Degree of corrosion (%) =
$$\frac{M_0 - M_f}{M_0}$$
 (2.8)

where M_0 is the initial mass of the electrode, and M_f is the final mass of the electrode after corrosion.

The material selection is challenging. Gold, silver and platinum are non-corrosive, but costly. Carbon is cheap, but it has extensive power consumption due to low conductivity (Mohamedelhassan and Shang, 2001). Although iron electrodes increase the flow of water up to twice the graphite electrodes (Segall and Brull, 1992), the precipitation of metal oxides increases the power consumption and, hence reduces the efficiency. Besides, the corrosion of steel and mild steel electrodes also decreases the effectiveness of soil stabilization (Lefebvre and Burnotte, 2002; Jayasekera and Hall, 2007).

Among metal electrodes, copper and stainless steel are known as the best options. In the case of using copper electrodes, the benefits are their low cost and availability, the oxygen consumption around electrodes, and the conductivity of copper oxide which reduces the voltage loss at the anode; however, a significant loss of power at the cathode was observed by Mohammedelhassan and Shang (2001). Stainless steel is more corrosion resistant and causes soil hardening by introducing iron to the soil as electrodes degrade (Liaki et al. 2010). But, it is not yet obvious that this reported soil strengthening, is due to the stainless steel degradation or reduction of pH below 5 due to which Al³⁺ dissolves from kaolinite. Stainless steel is corrosion resistant but generates chromate into the soil which is a source of contamination depending on the produced amount. According to Lockhart (1983) during soil dewatering, the type of the metal electrodes does not have an influence on the required amount of energy to achieve the targeted solid content.

An innovative material known as electrokinetic geosynthetics (EKGs) which is comprised of a conductive polymer was first developed in 1990 by researchers at the University of Newcastle in UK. Some of the other benefits of EKGs are their non-susceptibility to electrochemical reactions, providing filtration during drainage, and acting as a membrane. However, EKGs are not yet commercially available, and the connection between EKGs and the DC supply is not safe and sometimes results in disintegration of EKGs components which is not mentioned in the literature.

Electrode configuration

The configuration of electrodes affects the efficiency of the electrokinetic stabilization. There are two types of electrode configuration, one known as one-dimensional and the other as two-dimensional. A one-dimensional configuration is when for every anode one cathode is installed. The other type of electrode configuration is two-dimensional configurations in which there is more than one anode for a cathode. When this configuration is used, the acidic soil near the anode is extended to larger area due to the electrolysis. This is desirable when cementation near the cathode does not happen due to the presence of organic materials (Asavadorndeja and Glawe 2005). Depending on the type of chemicals in the soil, high pH develops near the cathode which causes soil cementation and increase the strength properties of the soil. However, if no cementation occurs, the increase in soil strength occurs mainly around the anode. In that case, the two dimensional electrode configuration is preferable.

2.7 Methods to improve electrokinetic stabilization

During electrokinetic stabilization due to the redox reaction, the environment near the anode becomes acidic which increases the corrosion rate of the anode. To reduce this effect on the anode, an alkaline chemical such as lime can be added from the anode to the soil. This method is referred as anode depolarization and improves the efficiency of the electrokinetic stabilization by reducing the corrosion rate of the anode. Another limitation of electrokinetic stabilization is the non-homogeneity of the soil sample between the anode and the cathode. This can be alleviated by reversing the polarity, referred to as polarity reversal. Also, the application of intermittent current has been proven to reduce the corrosion rate of the anode and consumption of energy during electrokinetic stabilization. Figure 2.4 gives a brief description on these methods to improve the limitations of the electrokinetic stabilization.

Polarity reversal

During electrokinetic stabilization, as water flows toward the cathode the soil near the anode starts to dry out and cracks may appear. This interrupts the electric current flow through the soil. For this reason, Shang et al. (1997) suggest polarity reversal. Polarity reversal and intermittent current reduce the corrosion of the anode, increase electroosmotic flow, and resulted in an increase of 570% in shear strength of the soil (Shang et al. 1997).

Due to the spatial changes in ionic concentration and fluctuation of pH, the treated soil is nonhomogenous and the strength near the anode is higher than near cathode due to the generation of negative pore pressure near the anode. The polarity reversal however improves the homogeneity of the treatment throughout the soil. Reversing the polarity in specific time intervals is much more effective than polarity reversal at the end of the electrokinetic stabilization. Because the soil has not yet lost its conductivity and the anode has not yet corroded completely, the polarity reversal works better when applied in time intervals. When metal electrodes are used, the polarity reversal is an option to reduce the effect of corrosion. The settlement rate of the soil decreases during polarity reversal (Bjerrum et al. 1967); however, the polarity reversal does not affect the amount of draining water (Ou et al. 2009); and higher energy is consumed during polarity reversal than in unidirectional current flow (Luo et al. 2005).



- Asavadorndeja and Glawe (2005) which ٠ shows that reversing the polarity can improve the efficiency by transporting hydrogen ions from cathode to anode performed a comparative study between normal and reversed polarity electrokinetic stabilization. He has reported an increase of shear strength by 570%.
- ٠ The polarity reversal can improve the distribution of the treatment throughout the soil. Therefore, it would be a good option to improve the evenness of the treatment by applying polarity reversal in time intervals or after completion of the electrokinetic stabilization.
- Other problems with this method are that ٠ the settlement rate of the soil decreases during polarity reversal (Bjerrum et al. 1967); the polarity reversal does not affect the amount of draining water (Ou et al. 2009); higher energy is consumed during polarity reversal than unidirectional method (Luo et al. 2005).

pH accelerates the corrosion rate of the anode, and decreases the precipitation of the pore fluid near the anode.

- The mobility of the hydrogen ions is much faster than hydroxides, that's why most part of the soil is acidic, and therefore the precipitation can only be limited near the cathode. To reduce this effect, the anode depolarization technique, which is firstly introduced, by Asavadorndeja and Glawe (2005) can be used.
- The strength of the soil can be increased in three cases where cations can be replaced, mineralization can occur and the pore fluid can be precipitates. The precipitations of the pore fluid can only occur when the pH of the soil is above 7 or only in alkaline environment. Therefore, it is desirable to keep the pH of the soil above 7. By addition of alkaline solution through the anode, the power consumption during electrokinetic can be reduced and the electric current distribution throughout the soil can be regulated.
- By addition of alkaline solution through the anode, the power consumption during electrokinetic can be reduces and the electric current distribution throughout the soil can be regulated.

and Kelsh (1975, 1980) which is the application of intermittent current. Intermittent current is an interruption of direct current within specific time intervals which can be regular or irregular. This method can reduce the rate of corrosion at anode and the overal power consumption without effecting the efficiency.

- ٠ Regular application of intermittent current is popular in the literature. It is shown that the application of intermittent current reduces the corrosion rate of the anode and the power consumption (Micic et al. 2001; Glendinning et al. 2008).
- Irregular intermittent current application is rarely reported in the literature.
- This technique is proposed by Sprute • and Kelsh (1975, 1980) which is the application of intermittent current. Intermittent current is an interruption of direct current within specific time intervals which can be regular or irregular. This method can reduce the rate of corrosion at anode and the overal power consumption without effecting the efficiency.

Figure 2.4 Methods to improve electrokinetic stabilization
Intermittent current

A common type of current used in the literature for soil stabilization is constant direct current. However, intermittent current appears to improve the electrokientic stabilization by reducing the corrosion rate at the anode (Sprute and Kelsh 1975, 1980; Shang et al. 1996). Intermittent current is an interruption of direct current within specific time intervals which can be regular or irregular. Micic et al. (2001) studied the effect of intermittent current on strengthening marine sediments and reported a reduction of power consumption by application of direct current applied for 2 minutes and turning off for 2 minutes. An increase of 185% in shear strength of the soil at the vicinity of the anode shows that the application of constant electric current (Micic et al. 2001).

When intemittent current is applied, the energy consumption is based on intermittent ratio which is determined as the ratio of the duration that electric current is applied t_{on} (h) over total treatment time t_{total} (h).

$$\tau = \frac{t_{on}}{t_{total}} \tag{2.9}$$

Energy consumption can be found from the following equation:

$$W = Pt_{total} \tag{2.10}$$

where W is the energy consumption (kWh/m³) and

$$P = \tau E_j \tag{2.11}$$

where P is power consumption per unit volume of soil (W/m³) which depends on electric field intensity ($E = -\frac{dV}{dl}$) with unit of V/m and current density of ($j = \frac{I}{A}$) with unit of A/m². The intermittent ratio (τ).

2.5 General laboratory designed electrokinetic models

Electrokinetic stabilization is not a conventional and standard method in geotechnical engineering and therefore does not have standard testing equipment or procedures. Researchers are required to develop their own instrument setups. The experimental set-ups presented in literature are classified as three models (Figure 2.5).

The first model is a plan view of the set-up and comprised of three parts: the electrokinetic cell, the soil, and the anode and cathode compartments that are referred as electrolyte chambers. The anode and cathode compartments can be designed as separate parts in the electrokinetic cell or can be as two hollow electrodes that are perforated from which the electrolyte solution is to be removed (Figure 2.5). The scheme with an electrolyte chamber is more popular in case of soil decontamination and using hollow metals as electrodes is a small-scale geotechnical simulation of electrokinetic field stabilization.

The second model is based on modification of triaxial or odeometer equipment. The concept of this model is same as model 1. This model has electrode chambers with a drainage at the cathode chamber. In this model, a surcharge can be applied to the soil using a loading plate. Application of surcharge improves the electrokinetic method. However, not only the use of this model of set-up is limited to laboratory scale experiments, this model is not applicable in case of soil slurry.

The third model is commonly used to dewater and stabilize slurries. This model is comprised of an electrokinetic cell with a settlement column that is connected to a plain Perspex sheet at the bottom. The rim of the settlement column is to be screwed to the plain Perspex sheet. The plain Perspex sheet has an opening in the centre to pass the wire from the electrode out of the system. The electrodes are to be placed at the bottom and top of the soil sediments with a filter paper in between. The process starts when the electrodes are connected to the direct current supply.

(Model 1)









Figure 2.5 Electrokinetic stabilization models presented in the literature

2.8 Theoretical framework for electrokinetic consolidation of soils

Numerous studies on electrokinetic stabilization either in-situ or in the laboratory have been performed (e.g. Gray 1970; Ozkan et al. 1999; Alshawabkeh and Sheahan 2002; Ou et al. 2009) with the ultimate conclusion that this method is one of the most innovative and cost-effective methods for treatment of different types of soils (especially clay and silty soils). However, there is still much to be investigated to optimize the effectiveness and efficiency of this treatment. Esrig (1968) has defined different boundary conditions based on the water flow through electrodes. For instance, if there is no water to flow from the anode (i.e. anode closed) the volume change by the electroosmosis consolidation method is equal to the volume of the draining water. Table 2.6 shows these boundary conditions and pore water pressure for possible cases of electroosmotic consolidation.

Electroosmotic permeability and electroosmotic flow adhere to the principle of Darcy's hydraulic law. The difference is that the electric gradient replaces the hydraulic gradient $\left(\frac{\Delta H}{\Lambda I}\right)$

$$Q_h = k_h i_h A = k_h \frac{\Delta H}{\Delta L} A \tag{2.12}$$

where Q_h (m³/s) is the quantity of water flow induced by hydraulic gradient in m³, A (m²) is the cross sectional area of the specimen through which the flow takes place (cm²), $\frac{\Delta H}{\Delta L}$ is hydraulic gradient where ΔH (*m*) is the hydraulic head and ΔL (*m*) is the length of the soil sample. Therefore, the electroosmotic theory of consolidation as proposed by Casagrande (1952) based on Darcy's law is:

$$Q_e = k_e i_e A = k_e \frac{\Delta V}{\Delta L} A \tag{2.13}$$

where Q_e (m³/sV) is the quantity of water in m³ which moves through an area of A (m²) under an applied gradient of $i_e = V/L$ (volt/m) to a soil with electroosmotic permeability of k_e (m²/V.s). The parameter, which governs the flow of water under an electric current, is electroosmotic permeability k_e , similar to k_h in conventional consolidation theory. According to Shang (1997), if both the anode and the cathode are open, then flow velocity can be determined by the following equation:

$$Q_e = k_e E \tag{2.14}$$

E (V/m) is the electric field intensity that is the electrical potential difference (V) over the distance d (m) and is the strength of the electric field. In fact, it is the force on a unit positive charge at a specific point.



 Table 2.6 Boundary conditions and pore water pressure (Modified from Esrig 1968)
 Page 1968



Theoretical framework for electrokinetic consolidation of soils

The electroosmotic consolidation theory is generated on the basis of consolidation theory, which is based on the quantification of water flow through the soil mass. The water flow in electroosmosis depends on the hydraulic conductivity (k_h) and electroosmotic permeability (k_e) of the soil. The flow velocity V_s (m/s) based on these two parameters is formulated by Schaad and Haefeli (1947). Following Schaad and Haefeli (1947) and Esrig (1968), Lewis and Humpheson (1974) formulated a two-dimensional flow based on two-dimensional application of electrical field (Figure 2.7).

All of the mentioned equations are derived in the following assumption: 1) the soil is saturated and homogeneous, 2) a constant electrical field with respect to time is applied, 3) there would be no electrochemical reactions, 4) there would be no movement of soil particles, 5) soil has uniform and constant physical property, and 6) all ranges of applied voltage is effective (Mitchel and Saga 2005). The electroosmotic formulations are mainly one-dimensional or two-dimensional with the similar concept.



Figure 2.7 Theoretical framework of electrokinetic stabilization

2.10 Summary

An overview of theoretical and experimental studies on electrokinetic stabilization has been presented in this chapter. The review shows the effect of different factors on the efficiency of electrokinetic consolidation. Understanding the effect of these factors on electrokinetic consolidation process helps to achieve optimal design criteria. Different methods have been reported in the literature to overcome the current limitations of electrokinetic stabilisation. However, further enhancement to the method should be adapted in the line of developing a standard design criteria and to make the method more accessible to geotechnical engineers.

Loss of soil-electrode contact, which is the result of oxidation and reduction reactions near the anode and cathode, interrupts the flow of electric current through the soil. This effect is mainly observed when metal electrodes are used. Electrokinetic geosynthetics (EKGs) seems to solve this problem. However, EKGs are not yet commercially available and the connection between the EKGs and power supply is not clearly defined in the literature, which seems to be a problem that needs further investigation before making any decision on using EKGs. Therefore, due to the low cost and availability of the metal electrodes, it is still desirable to use metal electrodes for laboratory and field simulations. There are very few studies on the effect of electrode material on electrokinetic stabilization of dredged mud in the literature. Therefore, the effect of electrode material needs to be further investigated.

After some time, the electric current applied to the soil drops due to an increase in soil resistivity, when the same voltage is applied. At this point, the applied voltage should be increased in order to maintain the electric current, which leads to higher power consumption. However, this is applicable when the electric current is intended to remain constant. To keep the electric current constant during the process, a stable direct current power supply required which is expensive. Effect of commonly available low cost power supplies is important to be investigated as it reduces the overall cost of the project. Very few studies investigated the effect of constant voltage and the literature is mainly focused on the constant electric current application. The effect of adopting a constant voltage (electric potential) needs to be further investigated.

Electrode configuration is another parameter that affects the efficiency of the electrokinetic stabilization. One-dimensional (using one cathode for each anode) and two-dimensional (using more than one cathode for each anode) electrode configurations are presented in the literature. One-dimensional electrode configuration is the most commonly used configuration for

stabilizing and strengthening soils, and it appears that a two-dimensional electrode configuration is rarely used. Very few studies have investigated the effect of electrode configuration on the electrokinetic stabilization of dredged marine sediments for land reclamation purposes. Therefore, in this study different electrode configuration are used to determine the best laboratory set-up to stabilize dredged marine sediments.

The electrokinetic stabilization set-up using the settlement column is presented in the literature. However, whether placing the anode on the top or at the bottom to maximize efficiency is not obvious. Further, a clear comparison based on soil settlement and its effect on electrokinetic stabilization is unavailable. Therefore, there is a need to investigate which electrode placement provides better results which is further investigated in this study.

As explained earlier in this chapter, methods such as application of intermittent current, polarity reversal and anode depolarization are potential methods to improve the drawbacks of the electrokinetic stabilization method. Application of intermittent current is reported to reduce the corrosion of the anode and improve the efficiency of the electrokinetic stabilization. However, this is true for application of regular intermittent current, which is mainly investigated in the literature. Investigations on effect of irregular application of electric potential on electrokinetic stabilization needs to be investigated.

All the theoretical models proposed in the literature for estimating the electrokinetic consolidation of soils are mainly known as Esrig (1968), Johnston and Butterfield (1977), Hu et al. (2012) models. The proposed model by Esrig is based on constant parameters such as electric potential and soil properties. However, in practice this model is far from reality since during electroosmotic stabilization many parameters are changing which results in improvement of the soil. In order to improve the Esrig model, Johnston and Butterfield (1977) proposed a model based on the changes of electric potential. But still both models were based on constant soil properties, thus further investigation is required to consider the changes of soil properties during electrokinetic stabilization. For this purpose, Hu et al. (2012) presented a model based on the changes of soil properties. However, the effects of physiochemical changes during electroosmotic consolidation are not investigated and this needs to be further examined.

In general, this chapter has reviewed the applicability and effectiveness of variety of methods and has discussed the limitations of electrokinetic stabilization as well as the different aspects of the method that needs further investigation.

Chapter 3: Materials and preparations

3.1 General

In this chapter, the properties of the soil and the method of sample preparation used throughout this study are discussed in detail.

3.1.1 Material: soil

Samples of dredged material used in this study are obtained from the land reclamation area in port of Brisbane, Queensland, Australia. When the sediments are dredged, they are often poured into the containment paddocks to go through self-weight sedimentation. When the dredged sediments are initially pumped into the paddocks, they have a very high water content ranging from 350% to 450%. The water content of the dredged marine sediments reduces to 250% to 300% due to air and sun drying. The dredged mud used in this study is taken from paddock C1 (Figure 3.1). This is due to two reasons, one is the moisture content of the sample at this paddock which is reduced to 250% and the other is its location since it is located at the centre of the area which is the ideal location of soil sampling as it is less disturbed by its surrounding.



Figure 3.1 Plan view of port of Brisbane showing the location from which the samples were taken

The mineralogy study of the dredged mud that is performed by Advanced Analytical Centre (AAC) at James Cook University is demonstrated in Table 3.1. The behaviour of the soil depends on the type of mineral present in the soil.

Name of the minerals	% weight	Importance **	Characteristics
		Mostly sand and	Retention of Na, k,
Quartz - (SiO ₂)*	31	silt	Ca, Mg, Fe(II), and
		Abundant	Silica, alkaline
Illite/ Muscovite - K ₂ [Si ₆ Al ₂]Al ₄ O ₂₀ (OH) ₄ *	21	mineral in clay	parent rock
		as a result of	Low pH, Removal of
		weathering	Na, K, Ca, Mg, Fe
			(II) and Silica can be
Kaolin - $[Si_4]Al_4O_{10}(OH)_8 \cdot nH_2O$ (n=0***			by leaching in fresh
or 4)*	25		water
		Easily	Low water and
Amphibole - (Ca, Na, K)2,3(Mg, Fe,		weathered and	humus content,
Al)5(OH)2	2	oxidized	limited leaching
Sodium Plagioclase - NaAlSi ₃ O ₈ *	-	Abundant	environment
Sodium Calcium Plagioclase - (Na,		mineral in not	
$Ca)Al(Si, Al)_3O_8*$	2	leached soils	
Potassium feldspar - KAlSi ₃ O ₈ *	11		
Calcite - (CaCo ₃)	-	Carbonate	
Halite - (NaCl) – (g/l)	35****	Salt	-
		Sulfide mineral	-
Pyrite - FeS ₂	5	- (Fool's gold)	
Expansive Clay - (Na, Ca)0.33(Al,		very soft	-
Mg) ₂ (Si ₄ O ₁₀)(OH) ₂ nH ₂ O*	2	phyllosilicate	
Organic Content (%)	0.4 - 1	-	-
*silicate groups, have soluble Na^+ , Mg^{2+} , K^+ , Ca^{2+} , Mn^{2+} , Fe^{2+} , and metal cations Co^{2+} , Cu^{2+} , and Zn^{2+} .			

Table 3.1 Mineralogy of the dredged mud obtained from port of Brisbane (modified from
Ganesalingam 2013)

** Sposito (2008)

*** n is the number of moles of hydration between layers

**** Laventure and Warkentin (1965)/ (Average organic content of the marine clays)

***** Quigley (1980)

The presence of metal elements in the soil especially contaminants affect its behaviour by interacting with its minerals and raise environmental issues. Pyrite (FeS₂) is one of the commonly found minerals from sulphides group. Pyrite is one of the unstable minerals that weather into sulphuric acid due to oxidation process and forms acid sulphate soils (Karathanasis, 1982). Pyrite has a very large surface area, and it reacts immediately in presence of air. Illite has a specific area of 65 to 100 m²/g with flaky particles. Dredged mud consist of

21% Illite. Silicate is the most abundant element on earth. Dredged mud is mainly consisting of Quartz and feldspars.

The ICP-AES (Inductively coupled plasma atomic emission spectroscopy that is used to detect the existence of trace metals) and ICP-MS (Inductively coupled plasma mass spectrometry is used to detect trace of metals at low concentration of one part in 10¹⁵ part per quadrillion) are conducted by Advanced Analytical Centre (AAC) at James Cook University. The metal elements and their concentration are presented in Table 3.2. According to the limits presented in environmental Australian Capital Territory (EPA 1999) and environmental protection (DEP 2004), the dredged material in this study is not contaminated.

Element	Concentration (mg/kg or ppm)	Element	Concentration (mg/kg or ppm)
Ag	<=0.05	Cu	18.1
Al	3970	Fe	28500
As	7.02	Hg	≤0.5
Ba	51.3	Mn	695
Be	0.9	Мо	1.02
Cd	0.431	Ni	17.2
Со	15.1	Pb	13.7
Cr	18.1	Sb	< = 0.05
Se	<=1	Ti	0.097
Zn	55.3	-	-

Table 3.2 Metal elements in the dredged mud (Ganesalingam 2013)

Table 3.3 shows the physical and chemical properties of the dredged mud along with the standard used to determine the related property. According to the Unified soil classification system, the dredged mud is classified as highly plastic sandy clay (CH).

Table 3.3 Physical properties of the dredged mud

Property	Value	Standard
Liquid limit (%)	92	AS1289.3.9
Plastic limit (%)	35	AS1289.3.2.1
Plasticity index (%)	57	PI = LL-PL

Linear shrinkage (%)	18-19	AS1289.3.4.1.
Sand (%)	20	
Silt (%)	38	AS1289 3.6.3
Clay (%)	42	
Specific Gravity	2.61	AS 1289.3.5.1.
Soil classification (USCS)	СН	Unified soil classification system
рН	8.05-8.13	
Electrical conductivity (mS/cm)	4.8	Refer to section.
Salt content (ppt)*	2.9	*ppt = Part per thousand
Colour	Grey	Visual identification
Activity	0.82-0.88	Activity = $\frac{PI}{\% Clay (particles < 2 \ \mu m)}$
Specific surface area (m ² /g)	130	Farrar and Coleman (1967) equation (\pm 20%) LL=19+0.56A _s where LL is liquid limit, and A _s is specific surface area in m2/g.

3.2 Methods

3.2.1 Sample preparation

3.2.1.1 Sample preparation for settlement columns

Step 1. setting the moisture content to the initial condition of dredged marine sediments: When dredged marine sediment is stored in the Geotechnical laboratory, its water content reduces to about 60% due to evaporation. Therefore, to replicate the initial condition of the soil received from the port, water is added to the soil to bring up the moisture content to 250% until consistent homogeneous slurry is obtained. The added water is either distilled water as a substitute for freshwater or saltwater that is taken from the seaside foreshore located in the Strand, Townsville. The salinity of the saltwater is measured using mass measurement and the correlation between salinity of the water at port of Brisbane and strand is determined (Figure

3.2). Accordingly, the saltwater from port of Brisbane has same salinity. The 0.0098% error is attributed to the laboratory condition and human error that may occur during the measurement. The correlation is determined by bringing up the salinity of the water to the set values of g/l concentration. For this purpose, 4 beakers are filled with a same specific amount of strand saltwater and 4 beakers are filled with saltwater from port of Brisbane. The concentration of the water is reduced by adding distilled water.

The correlation between the measured values of salinity with the probe in ppt and g/l is determined and shown in figure 3.2. The salinity of the saltwater is monitored using the probe shown in figure 3.3. To monitor the concentration of the saltwater while adding distilled water to reduce the salinity to the desired value, the probe is inserted to the saltwater samples. Once the equivalent salinity in ppt is reached the sample is left aside for salt content measurement using the mass measurement method. The method of mass measurement for determining the salinity of the water involves leaving the saltwater in the oven until completely dried, then the salinity of the saltwater is determined based on the measured mass of the salt crystals in proportion to saltwater mass that is tested.



Figure 3.2 Correlation between salinity of salinity obtained with probe and mass measurement a) Port of Brisbane saltwater b) Strand saltwater



Figure 3.3 Correlation between salinity of the saltwater from port of Brisbane and the salinity of the saltwater from Strand beach

Step 2. Removing stones and shells in the soil: when the water content of the slurry is brought to 250%, the soil mixtures are then passed through 4.25 mm sieve to remove any shell and large stones that interrupt the analysis.

Step 3: Measurement of soil salinity, electric conductivity and pH: when the soil is prepared the salinity of the slurry is determined with the following method:

After mixing the slurry thoroughly, couple of small representative samples are taken from the mixture. The samples are left in the oven to dry completely. The dried samples are then pulverised and passed through 2.36 mm sieve and mixed with water in 1:5 soil-water mass ratio. There are different methods of measurements. In traditional methods, soil samples are dried and ground before being brought up to 1:5 soil water ratio and, then, left for a specific time to settle (Allan and Hodgins 2003). The proposed waiting time before any measurements is 4-16 hours. Walker (2008) suggests a mixing time greater than 30 minutes and a waiting time of 15 minutes prior to any measurements. Rayment and Higginson (1992) proposed a mechanical shake of an hour and waiting time of 20-30 minute prior to measurements.

The ratio of soil and water at which electrical conductivity should be measured is another matter to be considered. The popular soil-water ratio for measurement of electrical conductivity and pH is 1:5 (dry mass) in Australia and 1:1 in America. In this study, 1:5 soil-water ratio is used for sample preparation. Then, the mixture is left on a magnetic stirrer for an hour to mix

thoroughly. Since dredged mud sediment settles slowly due to its low permeability, the mixtures are left to settle for 24 hours prior to any measurements. Then, the chemical properties of the mixtures are determined using a microprocessor-based waterproof pH/Conductivity/Temperature tester (Figure 3.4).

The electrical conductivity of the soil varies depending on soil porosity, water content, salinity, degree of saturation, and density. The conductance path is mostly through capillary liquid. The critical temperature at which electrical conductivity should be measured is 25 degrees Celsius. If the instrument does not have ATC (automated temperature correction), the measurements should be converted according to the following equation:

$$EC_{25} = f_t EC_t \tag{3.1}$$

where EC_{25} is the electrical conductivity of the soil measured at 25 degrees Celsius, EC_t is the electrical conductivity f_t is the temperature factor which can be determined from the following equation as suggested by Sheet and Hendricks (1995):

$$f_t = 0.4470 + 1.4034 \, e^{-t/26.815} \tag{3.2}$$



Figure 3.4 Waterproof pH/Conductivity/Temperature tester probe with calibration kit

Conclusion

This chapter explains the location where the samples for this study is taken from, the properties of the soil with its mineralogy, and the method of sample preparation to be treated with

electrokinetic stabilization. The method of measuring soil salinity is given. The salinity measurement is based on two methods of mass measurement and the probe measurement. The correlation between mass measurement and the probe measurement is determined to ensure that accuracy of the probe and to find a conversion factor between g/l and ppt. In next chapter, the effect of electrode configuration on electrokinetic stabilization is presented.

Chapter 4: Electrode configuration

4.1 Background

The configuration of electrodes has been reviewed in chapter 2. This chapter focuses on how different electrode configurations affect the electrokinetic stabilization of soils. The effectiveness of the process is largely a function of the extent of effective area where the soil changes physio-chemically, as it is these changes that produce electrokinetic stabilization. The physiochemical changes in the soil are due to the oxidation and reduction reactions that occur near the anode and cathode, respectively. The parallel installation of anodes and cathodes is the most common type of electrode configuration presented in the literature. This configuration was initially developed to decontaminate soils, and then was used by Casagrande in 1949 and later by other researchers to stabilize soils. With this configurations such as a two-dimensional configuration. As noted, it is the physiochemical reactions in the soil that result in cementation and soil strengthening, hence the broader the area affected by physiochemical reactions, the better the soil improvement will be.

A two-dimensional configuration is when more than one electrode of opposite polarity is installed for every electrode. Therefore, a greater area is affected by physiochemical reactions resulting in better stabilization. In a two-dimensional electrode configuration, if the purpose is specifically to improve the dewatering process, it is advisable to have more cathodes than anodes. However, if electrokinetic stabilization is being used specifically to improve the strength of the soil, anodes should be installed more dominantly. Sahib and Vinod (2010) showed that the tetrahedral configuration shape with the cathode at centre increased the shear strength of the soil by 76% and the tetrahedral configuration shape with the anode at the centre resulted in better dewatering by 33 %.

The effect of electrode configuration on performance reported in the literature, is mainly in relation to attempts to decontaminate soils (e.g. Kim et al. 2010; Cameselle et al. 2013; Jeon et al. 2015; Popescu et al. 2016). Some studies investigated the effect of electrode configuration on soil dewatering (Fourie et al. 2007; Rittirong et al. 2008; Sahib and Vinod 2010). However, very few studies investigated the effect of electrode configuration on electrokinetic consolidation and strengthening of dredged marine sediments. The effect of electrode configuration the effect of electrode configuration on dewatering and strengthening of dredged marine sediments is specifically the

topic of this study and is investigated and explained herein. The aim of this study is to investigate the best electrode configuration to maximize the electrokinetic settlement of the dredged mud.

4.2 Horizontal electrode configuration

4.2.1 One-dimensional and two-dimensional electrokinetic stabilization

A one-dimensional configuration is when the one-dimensional electromagnetic field is applied through identical number of anodes and cathodes that are installed in parallel to each other. In this case, the spacing between the electrodes governs the effectiveness of the treatment as it dictates the effective area that is treated. Figure 4.1a shows the effective area between the electrodes with different spacing in one-dimensional arrays. The spacing between electrodes affects the effectiveness of the electrokinetic stabilization. Shorter spacing results in better soil stabilization. However, a much shorter spacing requires a significantly larger number of installed electrodes. Installation of more electrodes results in higher electric density and as a result higher power consumption.

Two-dimensional electrode configurations increase the effectiveness of electrokinetic stabilization. Figure 4.1b shows the most popular two-dimensional electrokinetic stabilization arrangements. In this configuration, if number of anodes are greater than number of cathodes, the acidic environment created by electrolysis of the anode is extended to a larger effective area. This is desirable if cementation near the cathode does not happen. Cementation near the cathode occurs due to three reasons. The first reason is the accumulation of cations from metal electrodes that migrate from the anode to the cathode forming a cementing agent as a result of cation exchange reactions (Casagrande 1948). Second reason is the amount of organic matters in the soil. Soils with higher organic matter result in better cementation near the cathode (Ou et al. 2015). And the third reason is the level of soil alkalinity near the cathode resulting in cementing reactions, as cementation near the cathode occurs when the soil alkalinity is higher than 8.7. Depending on the soil mineralogy, the rise of pH near the cathode prepares the condition for soil cementation (Asavadorndeja and Glawe 2005). Cementation near the cathode results in an increase of soil strength. The strength of the soil near the anode increases due to the reduction of water content that is caused by electroosmosis. The electrolysis at the anode improves the strength of the soil through cation exchange reaction, which depends on the type of electrode used. The increase of strength near both electrodes results in better soil

improvement. However, if the soil near the cathode does not improve, the effective area that is improved can be increased by increasing the number of anodes if the purpose of the treatment is to improve the strength of the soil. Otherwise, for dewatering purposes installation of more cathodes result in better drainage.

The affected and non-affected areas are shown in Figure 4.1 for both one-dimensional and twodimensional electrokinetic stabilization. The coloured configuration is used in this study. The affected and non-affected areas are influenced by the electric field potential depending on the polarity of the electrodes. The area between two identical electrodes are not influenced by electric field since they repel each other. Only the area between two non-identical electrodes is affected by the electrokinetic stabilization. The electric field lines between two identical and non-identical electrodes are shown in Figure 4.2.



Figure 4.1 a) One-dimensional electrode configuration (2 array electrodes experimented in this study) b) Two-dimensional electrode configuration – 4 cathodes, one anode in the centre is experimented herein (Alshawabkeh et al. 1999b)



Figure 4.2 a) The electric field between two non-identical electrodes b) The electric field between two-identical electrodes (Ogborn et al. 2000)

In this section one-dimensional electrode configuration is adapted in the laboratory to investigate the effect of this type of configuration on electrokinetic stabilization of the dredged mud through monitoring the changes of consolidation characteristics.

4.3 One-dimensional electrokinetic stabilization trial

4.3.1 One-dimensional electrokinetic set-up

In this study to set-up the one-dimensional electrokinetic stabilization experiments, two hollow galvanized steel electrodes with inner diameter of 19 mm and outer diameter of 22 mm were used. The water was collected from the cathode due to electroosmosis. As explained earlier in chapter 2, electroosmosis is the flow of water from the anode to the cathode. Therefore, the water accumulates in the cathode, from where it can be collected. For this reason, the cathode is perforated and wrapped with a filter paper to avoid the blockage of the perforations. When the electrodes are placed in the Perspex box, the soil slurry that is prepared beforehand (for detailed sample preparation refer to section 3.2) is poured into the Perspex box until the sediment reaches a thickness of 70 mm. When the Perspex box is filled with the slurry, the electrodes are connected to the power supply and a voltage gradient of 30 Volts is applied to the slurry.

Figure 4.3 shows the schematic view of the one-dimensional electrokinetic set-up. This experiment is performed firstly to investigate the response of dredged marine sediments to electrokinetic stabilization and secondly to test the suitability of this electrode configuration. The effect of electrokinetic stabilization using two array electrodes is tested through examining its effect on consolidation properties of dredged marine sediments. To examine the consolidation behaviour of the dredged mud after electrokinetic stabilization, two samples of 38 mm diameter were taken. One sample was taken from the area near the anode and the other

sample was taken from the area near the cathode. These samples were then tested in an oedometer.



Figure 4.3 Schematic view of one-dimensional electrokinetic stabilization set-up

4.3.2 Compression and rebound index

Compression index or soil compressibility shows how much the soil settles due to consolidation. The degree of soil compressibility can be rated according to the compression index. Mitchell (1993) categorized soils as highly compressible if having a compression index greater than 0.4, moderate to intermediately compressible with compression index between 0.2 to 0.4, and low compressibility if it has a compression index less than 0.2. Table 4.1 shows the values of consolidation characteristics before and after electrokinetic stabilization that is obtained from the void ratio versus effective stress plot (Figure 4.4).

The compressibility of the same untreated dredged mud was determined by Ganesalingam (2011) as 0.693, which categorizes the dredged mud as a highly compressible soil. With the use of electrokinetic treatment, the compressibility of the dredged mud is reduced to 0.446 near the anode, and to 0.386 near the cathode, showing a significant improvement. The compressibility of the dredged mud reduced by up to a maximum 2.1 that of the untreated soil. The rebound index of the treated dredged mud, which is the slope of the rebound curve in void ratio-effective stress plot, near the anode and the cathode is 0.079 which is 88% lower than the rebound index of the untreated soil. Compression ratio which is $C_{c'}(1+e_0)$ reduced 4 times of the untreated dredged mud showing a reduction in plasticity index of the soil. However, the

recompression ratio $C_{r'}(1+e_0)$ increased showing an increase in proportion of recompression deformation.



Figure 4.4 Void ratio versus effective stress of the stabilized dredged mud

	Untreated Dredged mud*	Electro-kinetically treated dredged mud near the anode	Electro-kinetically treated dredged mud near the cathode
Compression index C _c	0.693	0.446	0.386
Rebound index C _r	0.078	0.079	0.079
Compression ratio CR	0.176	0.030	0.030
Recompression ratio RR	0.019	0.017	0.015

Table 4.1 Compression index, rebound index, compression and recompression ratio of the treated and untreated dredged mud

* Ganesalingam et al. (2011)

4.3.3 Coefficient of volume compressibility

Figure 4.5 shows the changes of coefficient of volume compressibility with changes of effective stress near the anode and the cathode. The m_v increased for $\sigma'_v < 12$ kPa and reduced as the effective stress increased, when the soil becomes stiffer. Similar behaviour was observed

in untreated samples with lower ranges of coefficient of volume compressibity and this is attributed to the difference between the initial void ratios (Ganesalingam et al. 2011).



Figure 4.5 Coefficient of volume compressibility versus effective stress

4.4 Two-dimensional electrokinetic stabilization trial

As explained in section 4.2, two-dimensional electrokinetic stabilization is when more than one opposite polarity electrode is installed for a single anode or cathode. The installation of electrodes can be in rectangular, hexagonal, or circular configurations with either the anode or the cathode in the middle. The choice of the middle electrode depends on the purpose of the experiment. If the purpose is to maximize the dewatering, the anode should be placed in the middle and if the purpose is to maximize the soil strength, then the cathode is to be placed in the middle. Herein, the anode is placed in the centre to maximize the dewatering of this dredged marine slurry which has an initial water content of 250%.

4.4.1 Two-dimensional electrokinetic Set-up

Five hollow galvanized steel electrodes were placed in a Perspex box ($300 \text{ mm} \times 200 \text{ mm} \times 200 \text{ mm}$), in a rectangular configuration as shown in Figure 4.6. The cathodes were wrapped with filer paper before placing them in the Perspex box to avoid the blockage of perforation. Once the slurry was poured into the Perspex box to a depth of 180 mm, the electrodes were then connected to the DC supply, and an electric potential of 30 V was applied.



Figure 4.6 Plan view of two-dimensional electrokinetic stabilization set-up

The undrained shear strength of the treated soil which gives an indication on improvement of soil strength properties, is measured using the laboratory vane (Figure 4.7). Vane shear strength of the samples were measured after completion of the test in different locations near the anode and the cathodes where samples were taken to measure the moisture content and chemical properties of the treated dredged mud. Figure 4.8 shows the measured undrained shear strength values and Figure 4.9 shows the water content of the treated dredged mud at the same location where the vane test was performed. It is seen that the shear strength near the cathodes was lower than at the anode, indicating that no significant cementation had occurred. However, the shear strength near the cathode was still much higher than the shear strength of the un-treated slurry due to the reduction of the moisture content.



Figure 4.7 The vane shear apparatus



Figure 4.8 Shear strength profile after electrokinetic stabilization



Figure 4.9 Water content profiles after electrokinetic stabilization

Figure 4.10 represents the relationship between the measured shear strength versus water content of the treated dredged mud. The relationship between shear strength and water content with 66% coefficient of correlation (r^2) is given as:

$$S_{\nu} = 65.54 \, e^{-0.027 \, w} \tag{4.1}$$

where S_u is the undrained shear strength (kPa) and w is the water content (%). This correlation can be used to determine the required increase in shear strength with reduction of water content of the treated dredged mud.



Figure 4.10 The relationship between water content and undrained shear strength

4.5 Vertical electrode configuration, investigation on placement of the anode and cathodes

Based on the one-dimensional and two-dimensional electrokinetic stabilization explained in sections 4.2 to 4.4, electrokinetic stabilization seems applicable and has potential to improve the consolidation and shear strength properties of dredged mud slurries. However, when the water content of the soil is greater than 1.5 its liquid limit, which is 2.7 times liquid limit in this study, the settling profile in horizontal (one-dimensional and two-dimensional) electrokinetic stabilization is not clear (Figure 4.11). This is due to the high surface area that leads to evaporation of the surface water and generation of desiccation cracks specifically near the anode. Therefore, to monitor the settlement of the dredged mud during the electrokinetic stabilization a vertical electrode configuration is implemented.

One of the key attributes of the vertical electrokinetic stabilization model is the placement of electrodes. The placement of electrodes is investigated to determine which method is better in maximizing the amount of settlement. Irrespective of where the anode is placed, the water

flows from the anode toward the cathode due to the electroosmosis process. However, if the anode is placed at the bottom, besides electroosmosis, soil particles also move toward the bottom of the column due to electrophoresis. In the next section the effect of electrode placement on the variation of electric current, electrokinetic settlement, and the efficiency of the method based on dewatering and cumulative power consumption are investigated.



Figure 4.11 Settling profile in one-dimensional and two-dimensional profile and generation of desiccation cracks

4.5.1 Vertical electrode configuration electrokinetic Set-up

Settlement columns made of plexiglass tubes were prepared with a height of 500 mm and internal diameter of 90 mm. The bottom of the column was attached to a plain Perspex glass sheet with a 3 mm diameter hole to admit the anode/cathode connection. Once the electrode, in the form of a circular disc, was placed on the plain Perspex sheet, the space around the 3 mm hole was sealed with silicone to avoid any leakage from the bottom and the four corners were screwed to the rim of the settlement column. Since the bottom of the column is closed, no drainage is allowed from the bottom. Therefore, the path of water flow is toward the top of the sediments providing a single drainage from the top only. When the drainage of water is not allowed from the bottom, the salinity of the entire soil sample can be controlled as the pore water accumulated on the top. However, in double drain condition the pore water dissipated

through the top and the bottom resulting in changes of concentration of salt in the treated soil. When the soil has high salt content the electric current varies and this variation of electric current results in an increase in power consumption due to an increase in soil resistivity (Jones et al. 2008).

Figure 4.12 shows the vertical electrokinetic set-up. The electrodes used are made of galvanized steel due to their availability and low cost. The electrodes are in the form of circular galvanized steel sheets where the anode has a thickness of 4 mm and the cathode has a thickness of 2 mm. The cathode is perforated to allow the drainage of water. Once the set-up is ready the slurry that is prepared with initial moisture content of 250% is poured into the columns in one instant (assuming that the lag time between pouring the slurry and starting the application of electric current is zero). A filter paper is located on the top of the dredged mud slurry and the electrode is placed on the top of the filter paper.

According to Hamed and Bhadra (1997) the same amount of dewatering with 80 % less energy consumption is obtained if the constant voltage is applied to the soil in increasing sequential order. Therefore, the electric potential is applied to the dredged mud slurry in two constant increments of 0 to 20 V and 20 to 30 V with gradient of 0.48 V/cm and 0.70 V/cm, respectively.



Figure 4.12 Vertical electrokinetic stabilization set up based on settlement column

4.5.2 Electroosmotic settlement

Figure 4.13 shows the changes in interface height (settlement) with time during electrokinetic stabilization. Higher settling rate, better sedimentation and flocculation of the particles were observed when the cathode is placed at the top. It can be seen that, when the anode is placed at the bottom the rate of settlement was much higher than when it is placed at the top. When the anode is at the bottom, the negatively charged particles migrate toward the anode to the bottom of the column due to the electrophoresis (movement of particles) and the pore fluid moves to the cathode due to the electroosmosis (movement of pore fluid). Therefore, more settlement was obtained when the cathode is placed at the top. Whereas, when the cathode is located at the bottom, negatively charged particles moved upward, which decelerated the settlement of the particle and resulted in generation of cracks within the soil.



Figure 4.13 Changes of interface height versus log time

When a slurry settles, it normally goes through three stages of sedimentation. First stage is the free settling stage at which particles settle individually. Second stage is hindered settling stage at which particles start to flocculate and settle together. Third stage is consolidation when the flocculated soil particles have sufficient grain contact between them with enough strength that the effective stress with the soil is greater than 1 kPa. These stages are identified from the time versus settlement graph when the trend of the data changes. When the cathode is at the top, the velocity of the hindered settling phase (that is the slope of the second portion of the settlement versus time), is much faster than when the cathode is at the bottom (Figure 4.13). This shows

that, the application of electric current to the soil accelerates the settling velocity. Therefore, electrokinetic stabilization is effective in increasing the free and hindered settling velocity of the soils and reducing the time of settlement (Mohamedelhassan and Shang 2001).

4.5.3 Variation of electric current with time

The current at the beginning of the test is a measure of the amount of generated ions in the soil, which presents as salt precipitates or as metal contaminants. Because the mobile ions are constantly electro-migrating toward the electrodes and the excess ions are neutralized by reacting with the soil the electric current reduces with time. H^+ and OH^- ions are generated at the electrodes as a result of redox reactions. Adsorption of these ions into the soil and the slow dissolution and precipitation of minerals and/or salt results in pH changes that lead to a long term and steady supply of charge carriers (Eykholt 1992). A constant voltage of 20 V has been applied to the soil at the beginning of the test. Once the electric current declined and approached zero, the voltage was increased to 30 V (at time t₃₀), on the assumption that the soil resistivity under application of a 20 V electric potential has reached its maximum and no further effect can be observed. However, once the voltage increased to 30 V, no significant changes in electric current was observed. This shows that further increase in voltage gradient did not affect the dredged sediment since it reached its maximum resistivity prior to application of further electric potential.

Figure 4.14 shows the variation of electric current with time for four cases where the cathode was placed at the top and bottom of the column, with the voltage applied being 20 V and then stepped up to 30 V. The variation in electric current with time follows a similar pattern regardless of anode/cathode configuration, however when the cathode was placed at the top, the maximum electric current was slightly higher and the dredged sediment experienced less peak resistivity. When the cathode was placed at the bottom of the settlement column lower electric current was observed when the voltage increases to 30 V. It is expected that as electrical resistivity of the soil increases the electric current reduces (Lo et al. 1991). However, in this study, the electric current of the soil increased and then reduced, implying an initial reduction of soil resistivity. The soil resistivity reduces until the electric current reaches a maximum of 0.55 Ampere.

Rittirong et al. (2008) used electrokinetic stabilization to improve the properties of soft clay soils. According to his study, when a voltage gradient of 25 V is applied, the electric current fluctuates at about 0.1 Ampere. Whereas, application of 45 V results in a reduction of electric current from 0.14 to 0.10 Amps after 6 days. The reduction in electric current is attributed to the voltage drop at the electrode-soil interface and the increase in resistivity of the electrodes. The electric current started to reduce when the resistivity of the anode was increased from 4 to 16 Ω /m after 6 days.

The reduction in electric current within the soil herein is solely attributed to the increase in resistivity of the soil. Since the drainage is from the surface of the sediment, water accumulates on the top. The accumulation of water on the top does not interrupt the flow of electric current through the soil as water is a good conductor of electricity. Higher voltage gradients result in higher soil resistivity and lower electroosmotic efficiency (Kuma 2005). When a voltage gradient is applied, the current flow increases showing an effective electroosmosis, however, at a specific point in time the electric current starts to fall due to the built-up resistance of the soil to the applied electric current (Karunaratne 2011). Lockhart (1983) argues that faster electroosmotic dewatering and higher final solid content with higher pH near the cathode are obtained with higher voltages. High pH results in depletion of water and higher resistance to the electric current flow. Therefore, the reduction of the electric current after 11 hours is attributed to an increase of pH near the cathode.



Figure 4.14 Variation of electric current with time

4.5.4 Efficiency of the method

The energy expenditure or power consumption defines the suitability and the efficiency of the method to dewater and consolidate the dredged sediment. Since the amount of energy consumption relates to the cost of electrokinetic stabilization, the efficiency of the method is evaluated against power consumption. The energy consumption is investigated based on measure of soil settlement, amount of dewatering and soil resistivity.

4.5.5 Settlement versus energy consumption

The total cumulative energy consumption when the cathode is at the bottom (C-Bottom) and when it is at the top (C-Top) are calculated as 136 kWh and 113 kWh, respectively. Therefore, it is desirable to place the cathode at the top for lower energy consumption, and higher settlement. High power consumption is often due to cracks in the soil surface and the generation of heat near the electrodes (Gray 1970). Asavadorndeja and Glawe (2005) reported the total cumulative energy expenditure of 120 kWh/m³ which is close to the energy consumed in this study to stabilize and dewater dredged mud slurries.

4.5.6 Dewatering versus energy consumption

Electroosmotic efficiency is defined as the quantity of water drained per unit of electrical current which is proportional to electroosmotic permeability (Jones et al. 2008). The reduction of water content and increase of shear strength is higher near the anode (Mitchell 1993). More reduction in water content results in higher dewatering efficiency. Dewatering efficiency is based on the amount of water to be removed in comparison with the initial moisture content (W_i) and defined as follow:

Dewatering efficiency =
$$(W_i - W_f)/W_i$$
 (4.2)

When water is drained and soil porosity reduces, the thickness of the electric double layer reduces such that it overlaps with that of the adjacent particle. The reduction in thickness of the double layer results in a reduction of zeta potential. The lower the zeta potential, the lower the hydraulic conductivity meaning that by application of the same voltage further settlement cannot be achieved. Therefore, higher voltages must be applied in order to further improve the dewatering. For this reason, the electric potential in the experiment were increased to 30 V after which a further reduction in water content is observed.

Due to the electromigration of the ions, most of the cations presented in the soil can be removed from the water that can be collected from the top where the cathode is located. Whereas, if the cathode is placed at the bottom, the contaminants in the form of cations are trapped in the sediment as there is no drainage at the bottom. Figure 4.15 shows that negatively charged ions (anions) and negatively charged particles settle while migrating toward the anode, and cations move toward the cathode in the direction of the water flow.



Figure 4.15 Electroosmotic and electrophoretic settlement of the dredged marine slurry

Figure 4.16 shows the amount of dewatering versus cumulative power consumption for both cases of anode-top-cathode-bottom and anode-bottom-cathode-top that was treated for duration of 9 days. Higher dewatering when the cathode was at the top is due to the complimentary benefits in the direction of electrophoresis and electroosmosis that occurred at the same time. However, when the cathode was at the bottom, the particles moved toward the anode. This slows down the process of sedimentation. The dewatering efficiency was calculated as 67% and 48% for the case where the cathode was placed at the top and where the cathode was placed at the bottom, respectively. Therefore, considering dewatering efficiency, it is more efficient to place the anode at the bottom and the cathode at the top.


Figure 4.16 Amount of dewatering with time during electrophoresis and electroosmotic settlement

4.5.7 Resistivity versus energy consumption

According to Acar et al. (1995) energy expenditure depends on the soil electrical conductivity, and the interface resistivity which is inversely proportional to the conductive areas between electrodes and soil (Zhuang and Wang 2007). As the soil electrical conductivity increases the energy required increases (Mohamedelhassan and Shang 2001).

Kuma (2005) evaluated the efficiency of the electrokinetic stabilization method through several field and laboratory trials. He showed that the efficiency of the method is based on power consumption and soil resistivity. Based on his study, the greater the surface area of the electrode, the lower the electrical resistance of the material, and better the efficiency of the method. The constant applied voltage over electric current varies as the interface height (volume of the treated soil) changes with time. Within a given time (t) and the electrical energy (E), the cumulative energy passing through the system is defined as:

$$E = V^2 t/R \tag{4.3}$$

By taking the logarithm of its components, equation (4.3) becomes

$$ln E = 2 ln V + ln t - ln R \tag{4.4}$$

If the voltage is constant, the cumulative energy consumption solely depends on the changes of electric current with time, which depends on changes of soil resistivity. Based on the field and laboratory studies, he defined a range of resistivity versus cumulative power consumption for which the electroosmotic stabilization can be effective, moderately effective or non-effective. Accordingly, the most effective electroosmotic stabilization is to be achieved if the ranges of electric current is greater than 10 A/m² and the ranges of soil resistance is less than 50 k Ω /m³, and if the ranges of electric current is less than 1 A/m² for the ranges of soil resistance greater than 500 k Ω /m³ the electroosmotic stabilization is not effective. Figure 4.17 shows the effectiveness of electroosmosis when the cathode is placed at the top (C-Top) and the bottom (C-Bottom). This figure compares the two arrangements of electrode placement, electroosmosis is effective for both cases. However, the cumulative energy is marginally lower when the cathode is placed on the top.



Figure 4.17 Efficiency of electroosmosis

4.6 Effect of electrokinetic stabilization on consolidation parameters with depth

In previous sections the effect of electrode configuration was investigated. It is shown that horizontal electrode configuration where the cathode is placed at the top is the most efficient design set-up to simulate the electrokinetic stabilization of dredged mud in the laboratory. It is also observed that the electrokinetic stabilization reduced the soils compressibility. In this section the effect of electrokinetic stabilization on compressibility of the dredged mud within depth is investigated. For this purpose, a constant electric current of 30 V is applied using horizontal electrode configuration explained in section 4.5.1. After electrokinetic stabilization, the treated sample is tested under oedometer consolidation.

When electrokinetic stabilization is completed, the strength of the treated dredged mud within 7 cm of the cathode is low. To ensure that the shear strength of the treated dredged mud near the cathode is enough for oedometer testing, a surcharge of 5 kPa is added to bring up the undrained shear strength near the cathode to 2 kPa, estimated using the following equation proposed by Skempton & Henkel (1953):

$$\frac{c_u}{\sigma'} = 0.11 + 0.0039(PI) \tag{4.6}$$

where c_u is undrained shear strength, σ' is the effective vertical stress (applied load), and *PI* is the plasticity index of the untreated dredged mud given in section 3.1.1 as 57%. Applying surcharge of 5 kPa is equivalent to applying 3.3 kg weight to the dredged mud. Since the dredged mud near the cathode is weak, the loading should be applied in stages. Therefore, everyday 0.3 kg is added through 10 days. After 10 days, the bottom rim of the column is opened and samples are taken from the bottom (near the anode which has higher undrained shear strength) to the top (near the cathode with lowest shear strength). Figure 4.18 shows how samples are taken from bottom of the column using an oedometer ring, and Figure 4.19 shows the location of the sampling.



Figure 4.18 Method of taking samples from bottom of the column



Figure 4.19 Location of the sampling with depth for oedometer testing

4.6.1 Compression index and Rebound index

Figure 4.20 shows the compression curves for treated dredged mud with depth. Sample A is taken from the area closest to the cathode, and sample C is taken from the area closest to the anode.



Figure 4.20 Compression curves of the treated dredged mud with depth

The compression index of the treated soil with depth is shown in Table 4.2. As discussed earlier, cementation near the cathode did not happen during stabilization of the dredged mud. Therefore, application of the surcharge increased the shear strength of the treated soil near the

cathode and increase the homogeneity of the treated soil sample. According to Mitchell (1993) if the compression index of the soil is less than 0.2, the soil has slight to low compressibity. This shows that the compressibity of the dredged mud has significantly improved with application of electrokinetic stabilization. The compression index and rebound index of the untreated dredged mud discussed in section 4.3.2 given as 0.69 and 0.078, respectively (Ganesalingam 2013). Showing a significant improvement in compressibity of the dredged mud when electrokinetic stabilization is used. The rebound index or recompression index of the treated dredged mud at all depths is equal to 0.163 showing an increase of 108% in comparison with the untreated soil that is 0.078 (Ganesalingam 2013). The soil compressibility of the treated dredged mud increased with depth.

Sample number	Compression index untreated dredged mud	Treated dredged mud	% reduction in compression index
А	0.69	0.57	21%
В	0.69	0.56	23%
С	0.69	0.52	32%

Table 4.2 Compression index of the treated dredged mud with depth

4.6.2 Pre-consolidation pressure

The preconsolidation pressure reduced from 25 kPa to 20 kPa from near the cathode to near the anode (Table 4.3). The value of the pre-consolidation pressure is determined using Casagrande graphical method. Pre-consolidation pressure is the maximum overburden pressure that is experienced by the soil in the past. The surcharge of 5 kPa is applied from the cathode. These values show that layer B and C is unaffected by the applied surcharge.

Table 4.3 Preconsolidation	pressure of	[•] the treated	dredged	mud with	n denth
	pressure of	me neureu	ul cuscu l		i acpin

Sample number	σ'_p (kPa)
А	25
В	20
С	20

4.7 Summary

In this chapter the effect of different electrode configurations is presented and discussed. Onedimensional and two-dimensional electrokinetic stabilization was used to investigate the potential application of electrokinetic stabilization to improve the properties of the dredged mud slurries. The experimental study on effect of one-dimensional electrode configuration showed that electrokinetic consolidation reduced the compression index of the dredged mud by 2.1 times and reduced the coefficient of volume compressibity, thus making the sediment stiffer. The recompression index is also reduced up to 88 % of the compression index of untreated dredged mud.

Generally, it was observed that the consolidation properties of the dredged mud improved by one-dimensional electrokinetic stabilization. However, when one-dimensional electrokinetic stabilization is used, the spacing between electrodes is an important design factor to be considered. The lower the spacing between electrodes, the better the effectiveness of the method. However, this results in higher power consumption. For more effective electrokinetic stabilization two-dimensional electrode configuration can be used. Two-dimensional electrokinetic stabilization improved the undrained shear strength of the dredged mud by reducing its water content especially near the anode. The water content and shear strength of the soil near the cathode did not improve as much as near the anode. This lowers the shear strength near the cathode, indicating the absence of organic matters which shows that no cementation had occurred.

Two-dimensional and one-dimensional electrokinetic stabilization can be used for the already existing lands and for natural soils with low hydraulic conductivity. More soil improvement can be obtained using two-dimensional electrokinetic stabilization. The installation of electrodes in the case of one and two-dimensional electrokinetic stabilization is not easy as the slurry does not have enough strength to hold the electrodes in place especially in land reclamation areas. In addition, the pull-out of the electrodes after electrokinetic stabilization affects the strength properties of the treated dredged mud. Therefore, the vertical electrode configuration was chosen to be used to investigate the electrokinetic stabilization parameters. However, the placement of electrodes was investigated to determine which electrode placement results in maximum dewatering and soil settlement. For this purpose, laboratory models where cathode placed at the top and at the bottom of the settlement column were evaluated based on

the resulting efficiency, power consumption, amount of dewatering, soil settlement, and soil resistivity.

Based on the result of the study, when 20 V electric potential is applied, the electric current increased till soil resistivity reached the maximum. The maximum soil resistivity is less, when the cathode is placed on the top. By applying additional electric potential, up to 30 V, further variations in electric current was not significant and electric current remained constant, showing that the soil has reached its maximum resistivity by applying an electric potential of 20 V. The dewatering efficiency improved from 48% (when cathode is placed at the bottom) to 67% (when cathode is located at the top) with less power consumption. Therefore, based on this experimental study, it is better to place the cathode on top to get maximum settlement and dewatering. This electrode configuration will be used throughout this thesis to investigate the effects of other design parameters on electrokinetic stabilization of the dredged mud sediments.

When cathode is placed on the top, the soil at deeper levels (near the anode) has more strength due to the reduction of water content as a result of electroosmosis and electrolysis of the anode. Therefore, the area near the cathode was improved by application of surcharge to increase the homogeneity of the treated dredged mud. It is shown that the compression index of the soil increased by 32%.

Chapter 5: Type of electric current application

5.1 General

In previous chapter the effect of electrode configuration was presented. It was concluded that the vertical electrode configuration is the most feasible and efficient laboratory configuration to obtain maximum settlement of the dredged mud and highest storage capacity. After determining the best electrode configuration for analysing the dewatering and settlement of the dredged mud slurries. The application of electric potential needs to be investigated to determine the voltage to be applied to the slurry during the electrokinetic stabilization.

Application of constant current is commonly used in the literature to stabilize soils. However, the electric current at which maximum settlement is to be achieved is not clear or standardized. In this study, different electric potential is applied for two main reasons 1) the application of constant current requires a constant electric power source, which is expensive and is counted as an additional cost to the project 2) to determine the optimum electric current at which maximum electrokinetic settlement can be obtained. For these purposes, an easily available laboratory power supply is used. This power supply works based on Ohms law:

$$V = IR \tag{5.1}$$

where if the voltage is kept constant the electric current (I) changes as the resistivity (R) of the soil changes. For different applied constant voltages, the variation of electric current with time which shows the resistivity behaviour of the soil is presented in this chapter.

Intermittent current is one of the methods that is reported in the literature to reduce the power consumption and therefore the cost of the project. Intermittent current is the application of electric current in on and off cycles, which can be regular or irregular. The regular application of intermittent current reported to reduce the corrosion of the anode beside power consumption (e.g. Micic et al. 2001). The regularity and irregularity of the intermittent current application can be determined from the intermittent current ratio. The intermittent current ratio is the ratio of the duration when electric current is applied to the soil over the total duration of the electrokinetic stabilization which includes the duration at which electric current is disrupted.

$$\tau = \frac{t_{on}}{t_{on} + t_{off}} \tag{5.2}$$

where t_{on} is the time during which the power is on (hours) and t_{off} is the time duration the power is off (hours). If the intermittent ratio is 0.5 then regular intermittent current is applied. If it is greater or less than 0.5 it means the application of intermittent current is irregular.

Buckland et al. (2000) showed that the constant application of electric current results in better sedimentation than regular intermittent current application with 50% intermittent ratio. However, there is a lack of research on the effect of irregular application of intermittent current, with intermittent ratio greater than 50%. Therefore, in this study application of constant electric current and irregular intermittent current of 70% intermittent ratio are investigated to determine if application of intermittent current with higher intermittent ratios would improve the efficiency of electrokinetic stabilization.

5.2 Effect of constant electric potential

Application of constant electric current for electrokinetic stabilization of soils is reported in the literature. If the electric current is kept constant, the electric potential changes as the resistivity of the soil increases. Same pattern is observed if the voltage is kept constant. In order to apply a constant electric current a special current source is required to limit the increase of electric current to the desired constant value. In that case, to bring up the electric current to the desired constant value. In that case, to be increases the timeframe of the electrokinetic stabilization as well as the cost needs to be invested for the direct current source. This process can be accelerated by applying a constant voltage (electric potential).

In this chapter, the effect of applied constant electric potential on settlement of the dredged mud slurry is investigated. For this purpose, electric potentials of different values are applied to determine how field intensity affects the electrokinetic dewatering and consolidation of the dredged mud. Series of electrokinetic set-up was prepared and the dredged mud slurry with initial moisture content of 250% is poured into the settlement columns. A constant electric potential of 2.5 V, 5 V, 10 V, 15 V, and 30 V is applied to dredged mud slurries to monitor the changes of electrokinetic sedimentation with different applied voltages. The maximum allowable electric potential (under safety regulations) that can be applied to the soil in the laboratory is 30 V. The effect of application of 20 V is given in chapter 4. To test the effect of low and higher ranges of voltage, the interval of 2.5 V is chosen for low voltages of 0-5 V and interval of 5V is chosen for 5-15 V.

It is concluded from previous chapter that electrokinetic settlement is maximized if vertical electrode configuration with the cathode on top is used. Therefore, the set-up used in this thesis is the set-up presented in chapter 4, section 4.5.

5.2.1 Variation of electric current with time

The variation of electric current with time, for the applied constant voltages, is shown in Figure 5.1. It is shown that in the first 40 hrs, the current in high voltage tests is significantly varied, especially for 30V test ($I_{max}/I_{min} \approx 5.5$). Meanwhile, the current in low voltage tests remains the same, showing the insignificant effect of electric potential on soil resistivity at low voltages up to 5V. The increase of electric current to the maximum in early stages of the process shows that the soil resistivity reduces initially and then increases after 40 hours.

The changes of electric current with time follows a similar pattern regardless of the voltage if the electric potential is greater than 2.5 V. Beside soil resistivity, the variation of electric current with time depends on the ionic concentration and movement of ions in the soil (Chou et al. 2004). When the electric current is at its highest value, the ionic concentration is at its maximum level. When the ions start to move and the concentration of ions reduces, the electric current reduces. The electric current reduces until the number of free ions are equal to the dissolved ions when the electric current remains constant.



Figure 5.1 Variation of electric current with time for different voltages

5.2.2 Electrokinetic sedimentation

The settlement of the dredged mud slurry with time during electrokinetic stabilization is shown in Figure 5.2. The free settling of the dredged mud finishes sooner when 30 V electric potential is applied, this is due to an increase in settling velocity of the soil. Electric potential has a direct relationship with settlement of the dredged mud. Therefore, the settlement of the dredged mud is higher if the applied electric potential is greater. When low electric potential of 2.5 V is applied, the settlement of the dredged mud is not completed within the timeframe of 20 days. However, when electric potential of 30 V is applied, the final settlement of the soil is maximum and the free settling and hindered settling of the dredged mud is completed within 20 days. Only when 2.5 V electric potential is applied, the settlement curve is not completed within 20 days.



Figure 5.2 Electrokinetic surface settlement of the dredged mud with application of different electric potential

5.2.3 Power consumption

The power consumption of the method depends on the electric current and the applied voltage. Higher electric current leads to higher power consumption of the electrokinetic stabilization. The Power consumption of the method is determined using the formula given in section 4.5. Cumulative energy consumption is the summation of the energy consumption at each time interval that is calculated with the measured electric current at that time interval.

$$W = \frac{u_0}{V} \int_0^t I(t) dt \tag{5.3}$$

Where u_0 is the applied voltage, I(t) is the electric current as a function of time, and V is the volume of the treated soil.

V (Volts)	I _{max} Maximum electric current (Amps)	Duration (hrs)	Cumulative energy consumption (kWh)
2.5	0.05	436	0.05
5.0	0.14	248	0.17
10	0.29	184	0.53
15	0.50	120	0.90
30	0.98	295	7.71

Table 5.1 Power consumption versus applied voltage for application of constant voltages

5.3 Effect of Irregular intermittent current application 5.3.1 Intermittent current electrokinetic Set-up

Figure 5.3 shows the intermittent current set-up. The vertical configuration where cathode is placed on top of the dredged mud slurry is used throughout this study. To investigate the effect of intermittent current, a timer is attached to the plug of the power supply to stop the electric current flow to the DC supply according to the program that is given to the timer. The timer is programmed to be turned on and off for intervals of 3 hours for 3 days. After 3 days, a 24 hours break is given to the system and the timer is reprogrammed for the next 3 days. To simulate the double drain and single drain sedimentation, a plain and a perforated sheet is screwed at the bottom to the rim of the settlement column.

5.3.2 Variation of electric current

Figure 5.4 shows the variation of electric current with time during electrokinetic stabilization for when electric current is applied to the soil. It is shown that when irregular intermittent electric current is applied, the electrical resistivity of the soil did not change within the first 4 hours. However, the electrical resistivity of the soil reduced, between 4 to 20 hours after which it started to increase. Two jumps are shown in the data, these jumps in electric current is due to the 24 hours in which the electric current is turned off. When constant electric current is applied, the resistivity of the soil started to increase from the beginning of the experiment until

80 hours after which it stayed constant. To analyse the electric resistivity of the soil the points at which electric current was zero is eliminated from the plot to be able to compare the trends of changes of electric current with time.



Figure 5.3 Set-up used to investigate the effect of intermittent current



Figure 5.4 Variation of electric current with time for double drained intermittent current (DDIC), single drained intermittent current (SDIC), and single drained constant voltage (SDCV)

5.3.3 Effect of Intermittent current application on soil sedimentation

The effect of providing drainage at the bottom of the column during electrokinetic sedimentation of the dredged mud is investigated along with the application of irregular intermittent current. It was observed that, the dissipation of pore pressure in the case of double drain condition is different to single drain in Terzaghi's theory of consolidation. However, in electrokinetic settlement the dissipation of pore water pressure in double drain condition was similar to the single drain behaviour. The similarity of the draining condition is due to the cementation near the anode that blocks the flow of water in double drain condition. Therefore, although a perforated sheet is provided at the bottom, the water did not drain out due to electroosmosis.

Figure 5.5 shows the changes of settlement with time. Intermittent current was applied to a singly drained (SDIC) and doubly drained sediments (DDIC). These two cases are compared to the single drained sediment when constant voltage is applied (SDCV). It is shown that the application of 70% irregular intermittent current did not change the sedimentation behaviour of dredged mud in comparison with application of constant voltage. Therefore, the time at which free settling and hindered settling of the dredged mud is completed was same for all cases. However, the final settlement of the dredged mud increased up to 4% when intermittent current is applied in comparison with constant voltage. The settlement of the soil did not change

if the water is allowed to flow from the bottom where simulating the double drain condition. Therefore, the electrokinetic settlement of both single and double drained sediments are almost identical.



Figure 5.5 Settlement versus time

5.3.4 Electroosmotic permeability

The electroosmotic permeability $(m^2/V.S)$ of soils is determined based on Darcy's law from the following equation given by Jones et al. (2011):

$$k_e = \frac{Q_e}{i_e * A} \tag{5.4}$$

where Q_e (m³/s) is the volume of the dewatered water per unit time in seconds, i_e is the gradient, and A (m²) is the cross-sectional area. The distance between electrodes is changing as soil settles since the cathode that is located on the top of the sediments moves downward toward the anode.

Figure 5.6 shows the relationship between electroosmotic conductivity and void ratio. It is shown that the electroosmotic permeability of the dredged mud increased with application of 70% intermittent current.



Figure 5.6 Electroosmotic conductivity versus void ratio relationships

5.3.5 Effect of drainage and intermittent current on efficiency of the method

Table 5.2 compares soil resistivity, power consumption, and the amount of dewatered water after electrokinetic stabilization of the dredged mud to determine the most efficient case. In case of the double drain condition the resistivity of the soil is much higher than the other two cases. This is attributed to the loss of water from the base at the beginning of the experiment which resulted in appearance of cracks at earlier stages of the experiment in comparison with other tests. Depending on the place of the sedimentation, dredged mud experiences desiccation even under self-weight sedimentation, and this occurs as soon as 30 days (Lee et al. 2013). With electrokinetic stabilization, the desiccation occurs sooner if the soil is properly drained out from the bottom or the top; and this interrupts the flow of electric current through the soil. Therefore, it is better to keep the water on the top to reduce the desiccation cracks throughout the soil which interrupts the flow of electric current within the soil specimen and increases the soil resistivity.

Earlier the effect of intermittent current on settlement is discussed, showing that the application of intermittent current increased the settlement of the dredged mud by 4% in comparison with application of constant voltage. When irregular intermittent electric current is applied the energy consumption is reduced to half of the application of constant voltage. If drainage is not

provided at the bottom of the column, the final resistivity of the soil is lower, this is due to the larger dry area near the anode that increased the soil resistivity in case of double drain condition.

Test ID	Final soil Resistivity (k û)	Cumulative power consumption (kWh)	Volume of the dewatered water (m ³)
SDCV	0.986	7.71	0.000133
SDIC	0.602	3.70	0.000149
DDIC	2.990	3.71	0.000153

Table 5.2 Effect of intermittent current on soil resistivity and power consumption

5.4 Summary

In this chapter, the effect of constant electric potential and its magnitude on electrokinetic settlement of the dredged mud is investigated along with the application of intermittent current and the effect of providing drainage. Following conclusions are drawn.

It is shown that the coefficient of electroosmotic permeability increased as the applied electric potential increased which led to a better soil sedimentation. However, increase in magnitude of the electric potential resulted in higher power consumption. When 30 V electric potential is applied to the soil, a significant increase in settlement was observed in comparison with application of lower voltages. Therefore, application of irregular intermittent current with 70% intermittent ratio studied and it is shown that intermitted current application resulted in higher settlement in comparison with application of constant electric potential of 30 V. Higher settlements were obtained while the energy consumption reduced to half. The power consumed by application of intermittent current for both double and single drained condition is almost the same.

When the drainage is provided from the bottom, the settlement or the process of electrokinetic consolidation did not significantly change in comparison with single drain condition, except increasing the generation of cracks in the soil specimen resulting in higher final resistivity. Therefore, intermittent current application using an impervious base and application of 70% irregular intermittent current is suggested to reduce the cost of the project.

Chapter 6: Effect of electrode material

6.1 General

The effect of electrode configuration and voltage gradient were discussed in previous chapters. The vertical configuration where the cathode is placed on the top is the most efficient configuration. The application of constant electric current is shown to be effective in increasing the settlement of the dredged mud slurry. As explained in chapter 5, higher electric current results in better sedimentation. Application of constant 30 V electric potential showed a significant improvement in soil settlement, therefore, to determine the effect of electrode material on optimum settlement of the dredged mud, 30 V electric potential is applied and the suitability of the electrode material is evaluated based on the corrosion rate and the improvement of the soil after electrokinetic stabilization.

In this chapter, the effect of electrode material is investigated and discussed. Different types of electrode material are used in the literature for electrokinetic stabilization of soils. However, a study on the effect of the different electrode materials on electrokinetic sedimentation is missing. For this purpose, the effect of electrode material on electrokinetic stabilization of dredged mud using aluminium, stainless steel, copper, and galvanized steel are investigated. The advantage, limitation, and the conductivity of these materials are given in Table 6.1.

Electrode Type	Advantageous	Limitation	Conductivity (S/m)
Copper	Lower wear rate, High resistant to DC current	Take longer to deburr a Copper electrode than to manufacture it. In addition to its high wear rate, it causes soil contamination and not environmentally friendly. Loss of voltage near cathode due to the high production of hydrogen gas. High Power loss at cathode	62*10 ⁶
Aluminium	High resistant to DC current, Light and Easy to shape, good conductor of electricity	Reacts with hydrogen ion (produce Aluminium-hydroxide) and cause compositional changes in the soil. Not environmentally friendly and a source of contamination.	37*10 ⁶
Steel, Mild steel, steel tubes	good conductor of electricity	High rate of Corrosion in presence of salinized soils is its main problem.	$16*10^{6}$

Table 6.1 Electrode materials used	l in this study to dewater and	l consolidate dredged marine sediments

Stainless steel	Easy to clean, Highly scratch resistance, Last long (high corrosion resistance)	Difficult to fabricate, not as malleable as other metals.	137*10 ⁶
Polyaniline coated electrodes: Polyaniline is a carbon based coating that increases the corrosion resistance of the metals.	Chemical compo	sition of the polyaniline: $-N \rightarrow N \rightarrow$	

6.1.1 Copper

During the electrokinetic stabilization, copper anodes are corroded significantly (Figure 6.1), due to oxidation with anions in discharged water, salinity of the soil and the tendency of copper to react and create copper oxide. Figure 6.1 shows the corroded anode after electrokinetic stabilization. An attempt to remove the anode resulted in breakage and disintegration of the anode.



Figure 6.1 The copper anode corroded such that it consumed and disintegrated completely

6.1.2 Aluminium mesh electrodes

When aluminium electrodes were used to stabilize the dredged mud, the anode and the cathode both corroded (Figure 6.2). The corrosion of anode is due to the electrolysis and the disintegration of cathode is attributed to the alkalinity of the adjacent area. The corrosion of the electrodes reduced the efficiency of the electrokinetic settlement. The dewatering of 37% did not make the slurry a workable material. Despite the reported studies in the literature on positive effects of aluminium electrodes on electrokinetic stabilization of soil (e.g. Liu et al. 2012), aluminium mesh is found non-effective in improving the properties of the dredged mud.



Figure 6.2 Corrosion of the aluminium anode (a) and aluminium cathode (b) after electrokinetic stabilization

6.1.3 Stainless steel electrodes

When stainless steel electrodes are used, the variation of electric current with time was more scattered than other metal electrodes (Figure 6.3). The electrical conductivity of the dredged mud near the cathode was approximately twice that of the untreated soil, which is the result of electromigration and production of hydroxides that increases the concentration of ions near the cathode. The dewatering and settlement of the dredged mud was successfully accomplished using stainless steel electrodes. However, scattered variation of electric current, and the related environmental concerns related to the generation of chromate, associated with the electrolysis of stainless steel coating, eliminates this metal from the electrode material options.



Figure 6.3 Variation of electric current with time

6.2 Galvanized steel and Polyaniline coated galvanized steel

As explained in section 6.1, corrosion of the anode is one of the criteria for choosing a proper anode. Aluminium mesh, copper and stainless steel electrodes were corroded during electrokinetic stabilization due to the reduction of pH near the anode. Another criterion for choosing a proper electrode is what enters the soil, for example stainless steel, copper, and aluminium are all source of contamination and not environmentally friendly. If these metals are used as the anode, care should be taken to control soil contamination by considering soil decontamination at the same time as soil stabilization. Steel that is coated with a layer of zinc oxide to reduce its corrosion rate is galvanized steel. Zinc is an element that is environmentally friendly and often used as a fertilizer for agricultural purposes. For this reason and due to its consistent electric current variations, galvanized steel is chosen as electrode material in this study.

Galvanized steel sheet with diameter of 900 mm and thickness of 3 mm is cut into circular plates (Figure 6.4) to be used as electrodes. Cathodes are perforated to allow the passage of water providing a drainage from the top and anodes are coated with polyaniline to investigate its effect on electrokinetic stabilization of the dredged mud. Polyaniline coating is a widely used conductive polymer that increases the aqueous corrosion resistance of metals (Santos et al. 1998). Polyaniline is obtained by electrochemical oxidation and polymerization of aniline (Skotheim and Reynolds 2007) which has unique properties such as low cost monomer,

electrochemical activity, stability, optical activity (Gvozdenovic et al. 2012). Since, polyaniline is effective in reducing the corrosion, especially in acidic environment (MacDiarmid and Ahmad 1997), it was selected as a coating material.



Figure 6.4 Circular galvanized steel

6.2.1 Variation of current density with time

The constant electric potential of 30 V or 5 V were applied to the dredged mud. When the electric potential was applied, the electric current was monitored for all cases until the tests were completed. Because electric current varied similarly in 30V coated and 30V non-coated tests, the fitting curves coincide at t_{opt} (Figure 6.5). Referring to the figure, when 5 V electric potential was applied to the soil through the coated anode, the pattern of changes in electric current with time was similar to when 30 V was applied through the coated anodes. This shows that the polyaniline coating increased the efficiency of the method by reducing the soil resistivity. However, this only happened when low voltage of 5 V was applied.

In case where electric potential of 30 V was applied through the coated anode, the soil resistivity was higher than the case of non-coated anode. This shows that the polyaniline coating was disintegrated at voltage of 30 V, resulting in higher resistivity and power consumption in comparison with non-coated anode. The disintegration of the polyaniline at voltage of 30 V will later be discussed in more detail through changes of physiochemical properties in next chapter.



Figure 6.5 Variation of electric current with time during electrokinetic stabilization of dredged mud slurry

As electric current and resistivity of the soil change with time and as soil particles settle, the density of the soil changes, and as a result the settling velocity varies. The variation of settling velocity with changes of electric current is given in Figure 6.6. The settling velocity of the soil is shown to be at its maximum when the electric current is the maximum. The detailed explanation on how settling velocity is obtained is given in section 6.2.3. The polynomial regressions to the measured settling velocities with respect to changes of electric current with application of 5 V and 30 V electric potential through coated and non-coated electrodes, are defined as:

When 30 V electric potential is applied through coated anode:

$$V_{\rm s} = -32.45 \, I^2 + 19.28 \, I + 0.53 \tag{6.1}$$

When 5 V electric potential is applied through coated anode

$$V_{\rm s} = -31.55 \, I^2 + 18.56 \, I + 0.564 \tag{6.2}$$

When 30 V electric potential is applied through non-coated anode

$$V_s = -27.14 \, I^2 + 5.22 \, I + 0.29 \tag{6.3}$$

When 30 V electric potential is applied through non-coated anode

$$V_{\rm s} = -4.09 \, I^2 + 5.79 \, I + 1.07 \tag{6.4}$$

where V_s is the settling velocity in mm/hr and *I* is the electric current in Amps. These regression equations are valid for electric current ranging from 0 to 1 amp showing that the electric current of the soil does not exceed 1 amp when 30 V electric potential is applied. The settling velocity based on electric current gives an information about soil resistivity. Since soil resistivity is a function of chemical changes in the soil, therefore it is necessary to investigate the relationship between settling velocity based on application of electric current. The maximum settling velocity is obtained at electric current of 0.3 amps in case of coated anode, and 0.6 amps in case of 30 V non-coated electrode, and less than 0.1 amps for when 5 V electric potential is applied.



Figure 6.6 Variation of electric current with settling velocity during electrokinetic stabilization of dredged mud slurry

The maximum electric current at optimum time and maximum electric current at maximum settling velocity that is obtained for each case is given in Table 6.2. When 30 V is applied through the non-coated anode, the maximum electric current is reached as soon as 8 hours. Whereas, if the anode is not coated, the maximum electric current with applying the same voltage is reached in 18 hours. When electric potential of 5 V is applied through the coated anode, same amount of time is required to reach the maximum electric current. This shows that

polyaniline coating improved the electrokinetic stabilization of the dredged mud if 5 V electric potential is applied and resulted in lower power consumption and higher electrokinetic efficiency.

Settling velocity is based on the hydraulic and electroosmotic permeability of the soil. Therefore, the relationship between electric current and settling velocity given in this section shows that the highest hydraulic and electroosmotic permeability was achieved at electric current of 0.3 amps when anodes are coated. Whereas, the ultimate current for non-coated anodes is twice the coated anode for application of 30 V electric potential and less than half for application of 5 V electric potential. Therefore, application of constant $I_{v(max)}$ is suggested electric current to obtain the maximum possible settlement. Use of $I_{v(max)}$ as the design electric current result in lowest power consumption and maximum settlement.

Electrode condition (anode)	V (volts)	Optimum time t _{opt} (Hours)	I _{max} Maximum electric current (Amps)	Iv _{s (max)} Electric current at maximum V _s (Amps)
Non-coated	5.00	90	0.20	<0.1
Coated	5.00	18	0.52	0.3
Non-coated	30.00	8	0.92	0.6
Coated	30.00	18	0.56	0.3

Table 6.2 Optimum time at which maximum electric current occurs

6.2.2 Electrokinetic sedimentation

The changes of interface height with time is shown in Figure 6.7. It is shown that if 5 V electric potential is applied through coated anode the sedimentation behaviour of the soil is similar to when 30 V constant electric potential is applied. Application of 30 V electric potential through coated anode did not significantly change the final settlement of the soil, and only 11% increase in final settlement of the dredged mud is observed. This is due to the disintegration of the polyaniline coating at higher voltage of 30 V, and is attributed to the unknown chemical reactions resulting from the interaction between polyaniline coating and dredged mud particles, or changes in resistivity of the anode by applying the polyaniline coating. However, when 5 V electric potential is applied through polyaniline coated anode, the final settlement of the dredged mud increased to 73% of the final settlement of the dredged mud in case of non-coated anode.

The logarithmic regression that is the best fit of the experimental data shows the settlement of the dredged mud at any time during electrokinetic stabilization given as:

When 5 V electric potential is applied through coated anode:

$$S = -47.38ln(t) + 472.84 \tag{6.5}$$

When 5 V electric potential is applied through non-coated anode

$$S = -31.521\ln(t) + 475.03 \tag{6.6}$$

When 30 V electric potential is applied through coated anode

$$S = -45.1\ln(t) - 463.69 \tag{6.7}$$

When 30 V electric potential is applied through non-coated anode

$$S = -38.37\ln(t) - 438.61 \tag{6.8}$$

where S is the settlement of the soil with time and t is the time interval in hours. These correlations are true for the dredged mud with the presented initial state and under the given laboratory condition.



Figure 6.7 Changes of settlement versus time

6.2.3 Electrokinetic sedimentation coefficient

The electrokinetic sedimentation coefficient is the slope of the porosity and settling velocity of the slurries. The sedimentation coefficient is related to the process of hindered settling prior to consolidation and it is similar to the coefficient of consolidation during the process of

$$r = \log_n(\frac{U}{V})$$

where *U* is the particle settling velocity (m/s):

$$U = \frac{g(\rho_s - \rho_f) d^2}{18 \,\mu}$$

where g is gravitational acceleration (m/s²), ρ_s and ρ_f are density of solid and fluid respectively (kg/m³), d is diameter of particles (m), and μ is the viscosity of fluid (N.s/m²). The coefficient of sedimentation can be determined from the plot of settling velocity and porosity. In this study the settling velocity and the porosity is determined using following equations:

$$n = \left(\frac{e}{1+e}\right) \times 100$$

where e is the void ratio, and n is the porosity in percent. The settling velocity (mm/hr) of the slurry based on changes of interface height is determined from:

$$V_s = \frac{\Delta H}{\Delta t}$$

where ΔH is the changes of interface height in (mm) within time interval of Δt (hr).

The variation of coefficient of sedimentation versus porosity is shown in Figure 6.8. The settling velocity increases as porosity increases up to 81% after which it reduces. The experiment starts initially with porosity of 86% which gradually reduces as sediment settles. In case of 5 V non-coated anode, the settling velocity did not change significantly. However, in case of 30 V non-coated, the settling velocity increases to a peak of 84% after which, it reduces progressively to a final porosity of 77%. Both cases of coated anodes follow a similar pattern. In both cases, the settling velocity starts at a relatively slow rate and increases progressively up to a maximum velocity, at porosity of 82%, then it reduces gradually reaching a final porosity of 74%.

The sedimentation coefficient increases when properties such as clay content of the soil increases (Shang 1997). The polyaniline contributes to changes in the properties of the dredged mud through unknown chemical reactions which result in better electrokinetic settlement and efficiency. The coefficient of electrokinetic sedimentation is calculated from the left hand side of the slope of the settling velocity versus porosity. The left side of the graph is related to the initial peak of the electric current, and first half of the settlement before consolidation, and the right hand side of the graph is considered as consolidation stage. Therefore, the coefficient of sedimentation is determined using the left hand side of the settling velocity versus porosity plot. The coefficient of sedimentation is 0.46 mm/hr, 0.98 mm/hr, 0.11 mm/hr, and 0.14 mm/hr for 5 V non-coated anode, 30 V non-coated anode, 5 V coated anode, and 30 V coated anode, respectively.

This shows that for coated anodes, the increase in voltage does not significantly change the settling velocity of the dredged mud. The electrokinetic sedimentation depends on the intrinsic properties of the clay and current density. The results reported in this study are in agreement with the findings of Shang (1997).



Figure 6.8 Variation of settling velocity with porosity

6.2.4 Electroosmotic permeability parameters

The soil settling velocity (given in section 6.2.1) is based on electroosmotic permeability of the soil. And the effectiveness of the electrokinetic dewatering depends on electroosmotic

permeability k_e . Electroosmotic permeability is measured using the equation given by Mitchell (1993):

$$q_{eo} = k_e A \frac{j}{\sigma} \tag{6.9}$$

where q_{eo} is the flow rate of water (m³/s), k_e is electroosmotic permeability m²/ (s.V), A is cross-sectional area between electrodes, j is current density in A/m², and σ is the electrical conductivity of the dredged mud slurry (S/m).

As dredged mud slurry settles, the electroosmotic flow reduces with time. The flow rate of water (q_{eo}) that is induced by application of electric current is measured using the volume of drained water within the time interval when interface height changes. Figure 6.7 shows the changes of electroosmotic permeability with void ratio. The electroosmotic permeability of the dredged mud reduced with reduction of void ratio. The responses of dredged mud to 30 V electric potential that was applied through coated and non-coated anodes were similar. Therefore, their fitting curves are nearly coincident. According to Malekzadeh and Sivakugan (2016), the electroosmotic permeability parameter which is assumed to be dependent on changes of interface height and index properties of the soil is given as:

$$k_e = \mathcal{C}^* e^{D^*} \tag{6.10}$$

where C^* and D^* are electroosmotic permeability parameters and e is void ratio. According to this equation the electroosmotic permeability parameters derived from the performed experiments are as follows:

- 5 volt non-coated anode: $C^* = 2 \times 10^{-6}$, $D^* = 2.45$ (6.11)
- 30 volt non-coated anode: $C^* = 7 \times 10^{-6}$, $D^* = 2.74$ (6.12)
- 5 volt coated anode: $C^* = 3 \times 10^{-5}$, $D^* = 1.77$ (6.13)

30 volt coated anode:
$$C^* = 7 \times 10^{-6}$$
, $D^* = 2.74$ (6.14)



Figure 6.9 Electroosmotic permeability versus void ratio

6.2.5 Power consumption

The power consumption in this study is based on two variables. One is the electric current which changes with changes of soil resistivity and time and the other variable is the distance between electrodes, since the cathode moves downward with as interface height changes. Therefore, the following equation is used to determine the consumption of energy.

$$P = \frac{dI}{dl} \times \frac{V}{A} \times \Delta t \tag{6.14}$$

where *P* is the power consumption (W.s/m³) to settle the volume of sediments noted as *dl* (changes in distance between electrodes), with a constant applied voltage of *V* (Volts), which generates an electric current difference of *dl* (Amps) within a time interval of Δt in hours through cross sectional area A (m²).

The energy consumption for treating 1 cubic meter of dredged mud when constant electric potential of 5 V and 30 V electric potential is applied to the soil through coated and non-coated anodes is shown in Table 6.3. The consumption of energy when a non-coated anode is used with a voltage of 5 V is 1.97 kWh/m³ which is very low in comparison with the other cases, however, the percent settlement is relatively lower than other cases too. When 5 V electric potential is applied to the dredged mud through the coated anode the power consumption

increases 1.8 times the case of non-coated anode. However, the coating increases the settlement up to 9.8 times. The power consumption increase when the anode is coated with polyaniline. This is attributed to the lower conductivity of polyaniline coating in comparison with the noncoated anode. The choice of whether to apply the polyaniline coating needs to be based on the budget and timeline of the project. The coated anode with application of 5 V constant electric potential is suggested as an effective anode for maximum electrokinetic efficiency.

Electrode condition (anode)	Energy consumption per 1 m ³ of dredged mud assuming power consumption has linear relationship with soil volume (kWh/m ³)	V (volts)	% settlement (ΔH/H _i)*100
Non-coated	1.97	5.00	28.75
Coated	19.34	5.00	50.00
Non-coated	71.88	30.00	42.50
Coated	112.73	30.00	47.50

Table 6.3 Power consumption, percent settlement and average electric current

6.3 Summary

In this chapter the effect of electrode material on electrokinetic stabilization of dredged mud is investigated. Different electrode materials such as aluminium, stainless steel, and copper are used as electrodes and it was observed that these metals corrode in short period of time. Therefore, galvanized steel electrode is chosen to be used as the electrode material in this study. However, galvanized steel corrodes in high saline environment. The galvanized steel anode is coated with polyaniline to reduce its rate of corrosion and to investigate its effect on electrokinetic sedimentation of the dredged mud.

Based on the results obtained, polyaniline coated anode increased the settlement of the dredged mud at low voltage of 5 V electric potential while increased the electrokinetic sedimentation coefficient from 0.46 to 0.98. Comparison between the changes of resistivity when coated electrodes are used shows that resistivity reduced when 5 V electric potential is applied, whereas it increased when 30 V electric potential is applied. The increase of resistivity during 30 V electric potential is due to the disintegration of polyaniline coating.

When 5 V electric potential is applied through the coated anode, although the power consumption is higher than non-coated anode, the settlement of the dredged mud is 1.8 times higher than when non-coated anode is used. Therefore, the application of 5 V electric potential with coated anode is beneficial and more economical, however the decision to use coated anode

should be based on the specific project features, including available budget and timeline. In this study, because 30 V electric potential with non-coated and coated anode shown a similar sedimentation to 5 V coated anode. The application of 30 V with non-coated anode is used for further investigations on other design parameters due to the lack of resources.

Chapter 7: Effect of electrode material on physiochemical properties

7.1 General

The electrokinetic stabilization affects the properties of soils by electrochemical reactions. The effects of electrode configuration and electrode material on electrokinetic sedimentation of the dredged mud are presented in chapters 3 and 4. However, little attention is paid to the physiochemical changes of the treated soil and their interpretation after electrokinetic stabilization. The physiochemical changes influence the soil behaviour by affecting the particles in microscale level. The changes of soil behaviour due to occurrence of the chemical reactions lead to changes in soil permeability, water filtration, strength, and soil erosion. Therefore, it is necessary to study the effect of electrokinetic stabilization on physiochemical behaviour of the dredged mud. For this purpose, samples at four different depths are taken from the treated soil (after electrokinetic stabilization is completed). Atterberg limits, changes in pH, electric conductivity, and salt content of the samples are measured. The changes of physiochemical properties of treated dredged mud when galvanized steel (DM-GS), aluminium (DM-AL), and coated galvanized steel (DM-CA) are used, are determined when all samples were treated under 30 V electric potential.

7.2 Changes of pH

Changes of pH are the consequences of electrochemical reaction during electrokinetic stabilization of soils. Specific chemical reaction only occurs at certain pH values. Soil pH affects the dispersion and flocculation of soil particles (e.g. Goldberg and Forster 1990; Chorover and Sposito 1995). The flocculation or dispersion behavior of soil is governed by salt content, cation valence, cation exchange capacity, and clay mineralogy (Goldberg et al., 1999). The cation exchange mechanism of illite, smectite, vermiculite, and hydroxides (goethite and hematite) is pH-dependant.

Clay particles have a negative surface charge and because of these charges the electrokinetic stabilization is efficient with soils that has clay content of greater than 30% (Mosavat et al. 2012). Zeta potential shows the amount of charges in the soil, and has a direct relationship with pH. The pH at which the soil is electrically neutral is known as point of zero net charge and is related to the pH of the electrolyte that the solid particle is submerged into. If the pH increases

above the point of zero net charge the negative charges increase, and the positive charges reduce. When soil is at pH of point of zero net charge, the soil particles start to flocculate quickly.

Figure 7.1 shows the changes of pH with depth after electrokinetic stabilization when aluminium (DM-AL), galvanized steel (DM-GS), and polyaniline coated (DM-CA) electrodes are used. The high pH (alkaline environment) near the cathode is due to the movement of H⁺ ions and the generation of hydroxides. Conversely, the pH near the anode is acidic, due to the electrolysis reaction. When anode is coated, the area near anode is more acidic than when galvanized steel and aluminium are used. However, the alkalinity of the soil near the cathode when coated anode is used is less than galvanized steel and greater than aluminium. When the pH is greater than 8.5 more hydrogen ions are dissolved in the soil and the surface charge is more negative (Baumgarten 2013). Showing that in all cases the soil near cathode has more negative surface charges.

The point of zero charge for magnesium oxide is less than 4, and for Fe or Al oxides it is greater than 7. The pH near the anode when galvanized steel and coated galvanized steel is used is less than 4. This shows the presence of magnesium oxides in the soil when galvanized steel is used, and the pH greater than 7 shows the presence of aluminium oxides when aluminium is used. When soil reaches the point of zero charge, this is an indication that the electrokinetic stabilization is completed. The low pH of less than 4.5 (Baumgarten 2013) in soils with higher salinity leads to a better aggregation and flocculation, and as a result the soil near the anode has higher strength in comparison with the other part of the treated sample.

Clay particles reorient and restructure themselves while settling. The strongest orientation occurs when soil pH reaches the neutral value of 7. This layer of soil is known as consolidation zone (Lewis 1967). The zone of consolidation in this study is recognized as a discoloration in the soil at a particular depth during electrokinetic stabilization (Figure 7.1b). When the changes of pH with depth is determined (the grey strip in Figure 7.1a shows the consolidation zone layer on the graph), it was observed that this layer has neutral pH, therefore, the consolidation zone layer in this study occurred at the depth between 13 cm to 15 cm. The depth at which consolidation zone layer occurs depends on the initial pH of the soil. The initial pH of the dredged mud (the yellow dashed line in Figure 7.1a) is 8.1. It was observed that the type of electrode material did not affect the depth at which consolidation zone layer happened.

In chapter 5, section 5.2 the variation of electric current with time is shown, and it is observed that the electric current approaches zero regardless of the applied electric potential. In chapter 4, section 4.5 it is shown that increase of electric potential from 20 V to 30 V did not improve the electrical resistivity of the soil. Citeau et al. (2011) studied the changes of pH with zeta potential and observed that at low pH of 2.6 the sludge is at its isoelectric point (zero point of charge). If the isoelectric point of dredged mud happens at low pH values, this can be another reason for interruption of the electric current.



Figure 7.1 a) Variation of pH with depth b) Zone of consolidation

7.3 Changes of electrical conductivity

The electric conductivity of the soil changes during and after electrokinetic stabilization due to the electromigration of ions. Figure 7.2 shows the changes of electric conductivity with depth. In general, the electric conductivity of the soil near the anode is higher than the electric conductivity near the cathode. The larger electric conductivity near the anode is due to the generation of conductive hydrogen ions and the lower electric conductivity near the cathode is due to the calcium hydroxide precipitation (Jayasekera and Hall 2007).

When hydrogen ions move toward the cathode due to electromigration, the electric conductivity of the soil increases proportionally as the concentration of H^+ increases. When the anode is coated with polyaniline the electric conductivity of the soil near anode is higher than using aluminium and galvanized steel electrodes, showing higher production of hydrogen ions at the anode when polyaniline coated electrodes are used. High electric conductivity near the anode shows the polyaniline coating is disintegrated when 30 V of electric potential is used.

It is observed from the plot, that the solubility of aluminium is higher than galvanized steel in electrokinetic stabilization. Therefore, the electrical conductivity of the soil near the anode when aluminium is used is less than when galvanized steel electrode is used. An increase in electric conductivity of the soil from anode to the cathode when galvanized steel is used shows the mobility of the magnesium in the soil. Magnesium can react with soil and cause soil hardening near the anode.

The electrical conductivity of the untreated soil is 4.8 mS (yellow dashed line in Figure 7.2). When galvanized steel is used, the electrical conductivity of the treated soil near the anode is lower than the electrical conductivity of the untreated dredged mud, showing that the magnesium that is released to the soil from galvanized steel is highly conductive and has migrated to the cathode.



Figure 7.2 Variation of electrical conductivity with depth

7.4 Changes of salt content

The hydrogen ion that is generated due to oxidation reaction near the anode, affects the electrical conductivity of the soil, therefore measurement of electrical conductivity is not a reliable measure of soil salinity (Jayasekera and Hall 2007). Therefore, in this study the salinity of the treated soil is measured directly using the probe presented in chapter 3. During electrokinetic stabilization dissolved salt in pore water migrates from the anode to the cathode. The movement of salt ions depends on their interaction with soil minerals.
During sedimentation, the clay particles repel each other due to their negative surface charges. Therefore, the settlement of clay particles takes a long time to be completed. In the presence of salt, the sodium and chloride ions neutralize the negative surface charges and accelerate the soil sedimentation. This is true for the self-weight consolidation. However, in case of electrokinetic stabilization, depending on salt concentration the rate of corrosion at the anode increases. As discussed earlier in chapter 2, the limit of salinity at which the electrokinetic stabilization starts to be inefficient is not clear in the literature.

Figure 7.3 shows the variation of salt content with depth. As shown in the Figure, in all cases the salt content near the anode is higher than the salt content near the cathode. In case of coated anode (DM-CA), the salt content of the dredged mud drops. When aluminium electrodes are used, the changes of salt content from anode to cathode are not as significant as when galvanized steel and coated anode is used. However, drop of soil salinity below 2.9 ppt from anode to the cathode shows that the salt content of the treated soil reduced when aluminium electrodes are used. Comparison between salinity of the treated soil when coated anode and galvanized steel is used, shows that higher desalination can be obtained using coated anode than galvanized steel.



Figure 7.3 Variation of salt content with depth

7.5 Changes of liquid limit

Changes of pH, electrical conductivity and the salt content alter the soil structure and as a result, its index properties. In electrokinetic stabilization, reduction of pH near the anode results in

lower liquid limits than near the cathode where an increase of pH causes an increase of liquid limit. However, the pH itself is not the only reason for changes in Liquid limit. The liquid limit of the soil mainly changes due to the alteration in soil mineralogy (Mitchell 1993).

Figure 7.4 shows the variation of liquid limit with depth. When aluminium electrodes are used, the liquid limit of the soil did not change significantly, although the pH of the soil changed abruptly from anode to the cathode. In case of galvanized steel and coated galvanized steel electrodes, the liquid limit of the soil reduced from the cathode to the anode. When galvanized steel is used, the liquid limit of the treated soil was smaller than original soil from depth 10 cm to 20 cm that is near the anode. This shows that the soil improvement is extended to a larger area when galvanized steel is used. Therefore, galvanized steel electrode is preferred for stabilizing the dredged mud sediments.



Figure 7.4 Changes of liquid limit with depth

7.6 Summary

In this chapter the effect of electrode material on physiochemical properties of the dredged mud is investigated using galvanized steel (DM-GS), coated galvanized steel (DM-CA), and aluminium electrodes (DM-AL). High electric conductivity near the coated anode, indicates that the coating is separated and entered to the soil at this voltage. It is observed that based on the changes of pH, electric conductivity, and salt content of the treated soil with depth, the galvanized steel and coated anode is preferable to aluminium electrodes. However, considering the changes of the liquid limit of the treated soil with depth when galvanized steel electrode is

used, the area of stabilization with lower liquid limit is extended to half of the depth showing an improvement in soil properties. Therefore, galvanized steel is preferable to coated galvanized steel.

Chapter 8: Summary, conclusion and recommendations

8.1 Summary

This chapter presents a summary of this thesis, conclusions and recommendations for future research. In general, this dissertation aimed to investigate the possibility to improve the properties of the dredged mud by changing the different design parameters and to maximize the efficiency of the method including power consumption, amount of dewatering, and electrokinetic settlement.

The overview of the electrokinetic stabilization method presented in chapter 2 shows the effect of different factors on the efficiency of electrokinetic consolidation. The effect of these factors on electrokinetic consolidation process is investigated in this thesis to achieve optimal design criteria and further enhancement to the method. One-dimensional, two-dimensional and vertical electrode configuration are studied. The comparison between electrode configurations shows that the efficiency of electrokinetic stabilization can be maximized by optimizing the distribution of electric field through changing the electrode configuration. A comparison based on soil settlement and its effect on electrokinetic stabilization between placing the cathode on the top and at the bottom is presented.

Type of electric current application is another factor that influences the electrokinetic stabilization due the changes of soil resistivity. The application of regular intermittent current is proven to positively improve the electrokinetic efficiency of the soil. The irregular intermittent current application is investigated as a possibility to further reduce the power consumption and cost of the treatment. Along with the application of intermittent current, the effect of providing drainage at the bottom of the settlement column is also studied. Electrodes often corrode during electrokinetic stabilization. The corrosion rate of the anode and its reactivity with soil depends on the type of the electrode material. The effects of electrode material on electrokinetic stabilization and the changes of physiochemical properties are presented along with the effect of polyaniline coating which is often used to reduce the corrosion rate of metals in acidic environment.

8.2 Conclusion

In this investigation, research has been carried out to determine the effect of various parameters on electrokinetic settlement of dredged mud. A summary of the pertinent findings can be found below.

- The experimental study on effect of one-dimensional electrode configuration showed that electrokinetic consolidation reduced the compression index of the dredged mud by 52 % and reduced the coefficient of volume compressibity, thus making the sediment stiffer. The recompression index is also reduced up to 88 % of the recompression index of untreated dredged mud.
- Generally, it was observed that the consolidation properties of the dredged mud improved due to one-dimensional electrokinetic stabilization. However, when one-dimensional electrokinetic stabilization is used, the spacing between electrodes is an important design factor to be considered. The lower the spacing between electrodes, the better the effectiveness of the method. However, this results in higher power consumption. For more effective electrokinetic stabilization two-dimensional electrode configuration can be used. Two-dimensional electrokinetic stabilization improved the undrained shear strength of the dredged mud by reducing its water content especially near the anode. The water content and shear strength of the soil near the cathode did not show significant improvement as near the anode.
- Based on the result of the study, when 20 V electric potential is applied, the electric current increased till soil resistivity reached the maximum. The maximum soil resistivity is less, when the cathode is placed on the top. When further electric potential of up to 30 V is applied the variations in electric current was not significant and electric current remained constant, showing that the soil has reached its maximum resistivity by applying an electric potential of 20 V. The dewatering efficiency improved from 48% (when cathode is placed at the bottom) to 67% (when cathode is located at the top) with less power consumption. Therefore, based on this experimental study, it is better to place the cathode on top to get maximum settlement and dewatering.
- The coefficient of electroosmotic permeability increased as the applied electric potential increased which led to a better soil sedimentation. However, increase in

magnitude of the electric potential resulted in higher power consumption. When 30 V electric potential is applied to the soil, a significant increase in settlement was observed in comparison with application of lower voltages. Therefore, application of irregular intermittent current with 70% intermittent ratio was studied and it is shown that intermittent current application resulted in higher settlement in comparison with application of 30 V. Higher settlements were obtained while the energy consumption reduced to half.

 Polyaniline coated anode increased the settlement of the dredged mud at low voltage of 5 V electric potential while increased the electrokinetic sedimentation coefficient from 0.46 to 0.98. Comparison between the changes of resistivity when coated electrodes are used shows that resistivity reduced when 5 V electric potential is applied, whereas it increased when 30 V electric potential is applied. The increase of resistivity during 30 V electric potential is due to the disintegration of polyaniline coating.

8.3 Further research

The findings of this study is to be used as the basis for further future research in electrokinetic dewatering and consolidation of dredged mud in land reclamation areas. The followings are recommended for further investigation:

- This study investigated the unknown effect of some design parameters on electrokinetic settlement of the dredged mud. The design parameters investigated in this study are influenced by changes of chemical properties of the soil such as changes of zeta potential. The effect of changes of zeta potential on electrokinetic settlement of the soils need to be further investigated. Further research is required to develop a correlation between zeta potential and the efficiency of the electrokinetic settlement.
- Further investigation on effect of intermittent current ratio greater than 50% on changes of electrokinetic settlement of the soil is desirable to reduce the cost of the method.
- When soil particles of the slurry settle, the density of the soil changes at each layer. The changes of density during the self-weight consolidation can be measured by sending an x-ray of the soil slurry. However, due to the safety issue only special laboratories are

allowed to perform this test in Australia. The undisturbed measurement of the density and pore pressure with depth is to be further investigated if possible.

- A comparative study of the laboratory investigation of electrokinetic stabilization and its field application need to be performed to test the implication of the proposed method.
- The simulation of electrokinetic settlement of the soil in which the estimation model of electrokinetic stabilization is determined, is required to estimate the settlement of the soil without laboratory investigation. However, extensive laboratory experiments on effect of chemical properties needs to be included in the numerical models.
- Homogeneity of the soil after the treatment and the generation of desiccation cracks can be investigated using unsaturated soil theories of Fredlund to investigate the effect of generation of crack on the settlement of the soil.

References

Abdullah, W. S., and Al-Abadi, A. M. (2010). "Cationic–electrokinetic Improvement of an Expansive Soil." *Applied Clay Science*, 47(3–4), 343-350. doi:http://dx.doi.org/10.1016/j.clay.2009.11.046.

Acar, Y. B., and Alshawabkeh, A. N. (1993). "Principles of Electrokinetic Remediation." *Environmental Science & Technology*, 27(13), 2638-2647. doi:10.1021/es00049a002.

Acar, Y. B., Gale R.J., Putnam G., and Hamed J. (1989). "Electrochemical Processing of Soils: Potential Use in Environmental Geotechnology and Significance of pH Gradients." *Paper presented at the 2nd International Symposiumon Environmental Geotechnology*, Envotech, Bethlehem.

Acar, Y. B., Gale, R. J., Alshawabkeh, A. N., Marks, R. E., Puppala, S., Bricka, M., & Parker, R. (1995). "Electrokinetic Remediation: Basics and Technology Status." *Journal of Hazardous Materials*, 40(2), 117-137. doi:<u>http://dx.doi.org/10.1016/0304-3894(94)00066-P.</u>

Acar, Y., Gale, R., Alshawabkeh, A., and Marks, R. (1995). "Electrokinetic Remediation: Basics and Technology Status." *Journal of Hazardous Materials*, 40(2), 117-137.

Allan, M., and Hodgins, L. (2003). "A faster laboratory method for testing soil electrical conductivity using a modified saturation extract procedure." *Communications in soil science and plant analysis*, 34(13-14), 1891-1896.

Alshawabkeh, A. N., Sheahan, T. C. (2002). "Stabilizing Fine-grained Sails by Phosphate Electrogrouting" *Geomaterials 2002: Soils, Geology, and Foundations. Washington: Transportation Research Board Natl Research Council.* 53-60.

Alshawabkeh, A., and Acar, Y. (1996). "Electrokinetic Remediation. II: Theoretical Model." *Journal of Geotechnical Engineering*, 122(3), 186-196. doi:doi:10.1061/(ASCE)0733-9410(1996)122:3(186).

Asavadorndeja, P., and Glawe, U. (2005). "Electrokinetic Strengthening of Soft Clay using the Anode Depolarization Method." *Bulletin of Engineering Geology and the Environment*, 64(3), 237-245. doi:10.1007/s10064-005-0276-7.

Azam, S., Jeeravipoolvarn, S., and Scott, J. (2009). "Numerical Modeling of Tailings Thickening for Improved Mine Waste Management." *Journal of Environmental Informatics*, 13(2), 111-118.

Azam, S., Scott, J. D., & Jeeravipoolvarn, S. (2007). "When Does a Slurry Become a Soil?" *Geotechnical News-Vancouver*, 25(3), 44.

Banerjee, S., and Mitchell, J. K. (1980). "In-situ Volume Changes Properties by Electroosmosis Theory." *Journal of Geotechnical and Geo-Environmental Engineering*, 106(ASCE15370).

Baumgarten, W., Dörner, J., and Horn, R. (2013). "Microstructural Development in Volcanic Ash Soils from South Chile." *Soil and Tillage Research*, 129, 48-60.

Bergado, D. T., and Sasanakul, I., and Horpibulsuk, S. (2003). "Electro-osmotic Consolidation of Soft Bangkok Clay using Copper and Carbon Electrodes with PVD."*American Society for Testing and Materials. West Conshohocken, PA, ETATS-UNIS,* 26.

Bergado, D. T., Balasubramaniam, A. S., Patawaran, M. A. B. &, & Kwuenpreuk, W. (2000). "Electroosmotic Consolidation of Soft Bangkok Clay with Prefabricated Vertical Drains." *Ground Improvement*, 4(4), 153–163.

Bjerrum, L., Moum, J. & Eide, O. (1967). "Application of Electro-osmosis to a Foundation Problem in a Norwegian Quick Clay." *Géotechnique*, 17(3), 214–235.

Buckland, D. G., Shang, J., and Mohamedelhassan, E. (2000). "Electrokinetic Sedimentation of Contaminated Welland River Sediment." *Canadian Geotechnical Journal*, 37(4), 735-747.

Burnotte, F., Lefebvre, G., and Grondin, G. (2004). "A Case Record of Electroosmotic Consolidation of Soft Clay with Improved Soil Electrode Contact." *Canadian Geotechnical Journal*, 41(6), 1038-1053.

Burton, P. (2010). *Pamela Burton Landscapes*. New York, NY, USA: Princeton Architectural Press. Casagrande, A. (1961). "Control of Seepage through Foundations and Abutments of Dams." *Geotechnique*, 11(3), 161-182.

Casagrande, L. (1948). "Electro-osmosis." *Proceedings 2nd International Conference on Soil Mechanics and Foundation Engineering*, Rotterdam, 218–223.

Casagrande, L. (1949). "Electro-Osmosis in Soils." *Géotechnique*, 1, 159-177. Retrieved from http://www.icevirtuallibrary.com/content/article/10.1680/geot.1949.1.3.159.

Casagrande, L. (1952). "Electro-osmosis Stabilization of Soils." *Journal of the Boston Society of Civil Engineers*, 39(1), 51–83.

Chappell, B. A., and Burton, P. L. (1975). "Electro-osmosis Applied to Unstable Embankment." *Journal of the geotechnical engineering division*, 101(8), 733-740.

Chen, K. J., and Jia, P. (2007). "A Genetic Algorithm for Dynamic Advanced Planning and Scheduling (DAPS) with a Frozen Interval." *Expert systems with Applications*, 33(4), 1004-1010.

Chien S.C., Ou C.Y., Wang Y.G. (2009). "Injection of Saline Solutions to Improve the Electro-osmotic Pressure and Consolidation of Foundation Soil." *Applied Clay Science*, *44*(3-4), 218-224.

Chorover, J., and Sposito, G. (1995). "Colloid Chemistry of Kaolinitic Tropical Soils." *Soil Science Society of America Journal*, 59(6), 1558-1564.

Chou, S. B., Cheng, M. C., and Yen, S. C. (2004). "The Enhanced Removal of Cadmium and Lead from Contaminated Soils and the PH effect by Electrochemical Treatment." *Journal of Environmental Science and Health, Part A*, 39(5), 1213-1232.

Citeau, M., Larue, O., and Vorobiev, E. (2011). "Influence of Salt, pH and Polyelectrolyte on the Pressure Electro-dewatering of Sewage Sludge." *Water research*, *45*(6), 2167-2180.

El Naggar, M. H., & Routledge, S. A. (2004). "Effect of electro-osmotic treatment on piles." *Proceedings of the Institution of Civil Engineers-Ground Improvement*, 8(1), 17-31.

Esrig, M. I. (1968). "Pore Pressures, Consolidation and Electro-kinetics." ASCE Journal of the Soil Mechanics and Foundations, 94(SM4), 899-921.

Esrig, M. I., and Henkel, D. J. (1968). *Some Aspects of Soil Mechanics in Relation to Vehicle Mobility*. Technology & Engineering, Cornell University, Retrieved from Buffalo Inc.

Eykholt, G. R. (1992). "Driving and complicating features of the electrokinetic treatment of contaminated soils." PhD dissertation. University of Texas at Austin.

Fourie, A. B., & Jones, C. J. F. P. (2010). "Improved Estimates of Power Consumption During Dewatering of Mine Tailings using Electrokinetic Geosynthetics (EKGs)." *Geotextiles and Geomembranes*, 28(2), 181-190. doi:http://dx.doi.org/10.1016/j.geotexmem.2009.10.007.

Fourie, A. B., Johns D.G., and Jones C.J.F.P. (2007). "Dewatering of Mine Tailings using Electrokinetic Geosynthetics." *Canadian Geotechnical Journal*, 44(2), 160-172.

Ganesalingam, D., Arulrajah, A., Ameratunga, J., Boyle, P., and Sivakugan, N. (2011). "Geotechnical properties of reconstituted dredged mud." *Proc.14th Pan-Am conference, Geotechnical conference,* Toronto, Canada.

Ganesalingam, Dhanya. (2013). "Consolidation properties of recent dredged mud sediment and insights into the consolidation analysis." PhD thesis, James Cook University.

Glendinning, S., Jones, C. J. F. P., Lamont-Black, J., & Hall, J. (2008). "Treatment of Lagooned Sewage Sludge in Situ Using Electrokinetic Geosynthetics." *Geosynthetics International*, 15(3), 192-204. doi:10.1680/gein.2008.15.3.192.

Glendinning, S., Jones, C.J.F.P., Pugh, R.C. (2005). "Reinforced Soil using Cohesive Fill and Electrokinetic Geosynthetics (EKG)." *International Journal of Geomechanics*, 5(2), 138-146.

Goldberg, D. E., Rajaniemi, T., Gurevitch, J., & Stewart-Oaten, A. (1999). "Empirical Approaches to Quantifying Interaction Intensity: Competition and Facilitation Along Productivity Gradients." *Ecology*, *80*(4), 1118-1131.

Goldberg, S., and Foster H. S. (1990). "Flocculation of reference clays." *Soil Science Society of America Journal*, 54, 714-718.

Gray, D. (1970). "Electrochemical Hardening of Clay Soils." Geotechnique, 20(1), 81-93.

Gvozdenovic, M. M., Jugovic, B. Z., Trisovic, T. L. J., Stevanovic, J. S., Plazinic, M. V., and Grgur, B. N. (2012). "Electrochemical Characteristics of Polyaniline Lead Dioxide Powder." *Annals of the Faculty of Engineering Hunedoara*, 10(3), 445.

Hamed, J. T., & Bhadra, A. (1997). "Influence of Current Density and pH on Electrokinetics." *Journal of Hazardous materials*, 55(1), 279-294.

Hamed, J., Acar, Y., & Gale, R. (1991). "Pb(II) Removal from Kaolinite by Electrokinetics." *Journal of Geotechnical Engineering*, 117(2), 241-271. doi:doi:10.1061/(ASCE)0733-9410(1991)117:2(241).

Hamir, R. B., Jones, C., & Clarke, B. G. (2001). "Electrically Conductive Geosynthetics for Consolidation and Reinforced Soil." *Geotextiles and Geomembranes*, 19(8), 455-482. doi:10.1016/s0266-1144(01)00021-8.

Heister, K., Keijzer, T. J. S., and Loch, J. P. G. (2004). "Stability of Clay Membranes in Chemical Osmosis." *Soil Science*, 169(9), 632-639. doi:10.1097/01.ss.0000142635.69858.1b.

Helmholtz, H. L. F. (1879). "Studies of Electric Boundary Layers." Wied. Ann., 7, 337-382.

Ho, M. H., and Chan, C. M. (2011). "Some Mechanical Properties of Cement Stabilized Malaysian Soft Clay." *World Academy of Science, Engineering and Technology*, 74, 24-31.

Horseman, N. J., Bassom, A. P., & Blennerhassett, P. J. (1996). "Strongly Nonlinear Vortices in the Stokes Layer on an Oscillating Cylinder." *Proceedings: Mathematical, Physical and Engineering Sciences*, 452(1948), 1087-1111. doi:10.2307/52891.

Hu L. M. Wu W. L. and Wu, H. (2012). "Numerical Model of Electroosmotic Consolidation in Clay." *Geotechnique*. 62(6), 537-541.

Indraratna, B., Lackenby, J., and Christie, D. (2005). "Effect of Confining Pressure on the Degradation of Ballast under Cyclic Loading." *Geotechnique*, 55 (4), 325–328.

Jayasekera, S., and Hall, S. (2007). "Modification of the Properties of Salt Affected Soils using Electrochemical Treatments." *Geotechnical and Geological Engineering*, 25(1), 1-10. doi:10.1007/s10706-006-0001-8.

Jeyakanthan, V., Gnanendran, C. T., and Lo, S. C. R. (2011). "Laboratory Assessment of Electroosmotic Stabilization of Soft Clay." *Canadian Geotechnical Journal*, 48(12), 1788-1802. doi:10.1139/t11-073.

Johnston, I. W., Butterfield, R. (1977). "A Laboratory investigation of soil consolidation by electroosmosis." *Australian Geomechanics Journal*, 21-32.

Jones, C. J. F. P., Glendinning, S., Huntley, D. T., and Lamont-Black, J. (2006). "Soil consolidation and strengthening using electrokinetic geosynthetics—concepts and analysis." *Geosynthetics*. Millpress, Rotterdam. Jones, C. J. F. P., Lamont-Black, J., Glendinning, S., Bergado, D., Eng, T., Fourie, A., and Yan-Feng, Z. (2008). "Recent research and applications in the use of electrokinetic geosynthetics." *Proceedings of 4th European geosynthetics conference*. EuroGeo4, Keynote paper: Edinburgh, UK.

Jones, C. J., Lamont-Black, J., and Glendinning, S. (2011). "Electrokinetic Geosynthetics in Hydraulic Applications." *Geotextiles and Geomembranes*, 29(4), 381-390.

Kalumba, D., Glendinning, S., Rogers, C., Tyrer, M., and Boardman, D. (2009). "Dewatering of Tunneling Slurry Waste Using Electrokinetic Geosynthetics." *Journal of Environmental Engineering*, 135(11), 1227-1236. doi:doi:10.1061/(ASCE)0733-9372(2009)135:11(1227).

Kamarudin, B., Mohd. R. T., Khairul, A. K. (2010). "Electrokinetic treatment on a tropical residual soil." Paper presented at the Proceedings of the ICE - Ground Improvement.

Kaniraj, S., Huong, H. L., and Yee, J. H. S. (2011). "Electro-Osmotic Consolidation Studies on Peat and Clayey Silt Using Electric Vertical Drain." *Geotechnical and Geological Engineering*, 29(3), 277-295. doi:10.1007/s10706-010-9375-8.

Karunaratne, G. (2011). "Prefabricated and Electrical Vertical Drains for Consolidation of Soft Clay." *Geotextiles and Geomembranes*, 29(4), 391–401-391–401.

Kim, B. K., Baek, K., Ko, S. H., & Yang, J. W. (2011). "Research and Field Experiences on Electrokinetic Remediation in South Korea." *Separation and purification technology*, 79(2), 116-123.

Kuma, J.V.M., (2005). "Optimized electrically conductive vertical drains for electro-osmosis treatment of soft soils." Ph.D. Thesis, National University of Singapore, Singapore.

Lee, B. Y., Kim, Y. Y., Yi, S. T., and Kim, J. K. (2013). "Automated Image Processing Technique for Detecting and Analyzing Concrete Surface Cracks." *Structure and Infrastructure Engineering*, 9(6), 567-577.

Lefebvre, G., and Burnotte, F. (2002). "Improvements of Electroosmotic Consolidation of Soft Clays by Minimizing Power Loss at Electrodes." *Canadian Geotechnical Journal*, 39(2), 399-408.

Lewis, D., and Quirk, J. (1967). "Phosphate Diffusion in Soil and Uptake by Plants." *Plant and Soil*, 26(1), 99-118.

Lewis, R. W., & Humpheson, C. (1974). "Numerical Analysis of Electro-osmotic Flow in Soils." *Journal of the Soil Mechanics and Foundations, ASCE,* 11(3), 51. doi:http://dx.doi.org/10.1016/0148-9062(74)91583-6.

Liaki, C., Rogers, C. D. F., & Boardman, D. I. (2010). "Physico-chemical Effects on Clay due to Electromigration using Stainless Steel Electrodes." *Journal of Applied Electrochemistry*, 40(6), 1225-1237. doi:10.1007/s10800-010-0096-8.

Liu, M., Yin, X., and Zhang, X. (2012). "Double-layer Graphene Optical Modulator." *Nano letters,* 12(3), 1482-1485.

Lo, K. Y., Inculet, I. I., & Ho, K. S. (1991). "Electroosmotic strengthening of soft sensitive clays." *Canadian Geotechnical Journal*, 28(1), 62-73.

Lo, K. Y., Inculet, I. I., and Ho, K. S. (1991). "Electroosmotic Strengthing of Soft Sensitive Clays." *Canadian Geotechnical Journal*, 28(1), 62-73.

Loch, J. P. G., Lima, A., & Kleingeld, P. (2010). "Geochemical Effects of Electro-osmosis in clays." *Journal of Applied Electrochemistry*, 40(6), 1249-1254. doi:10.1007/s10800-010-0098-6.

Lockhart, N. (1983). "Electro-osmotic Dewatering of Clays: III. Influence of clay type, exchangeable cations, and electrode materials." *Colloid and Surfaces*, 6, 253-259.

Lockhart, N. C. (1983). "Electroosmotic Dewatering of Clays: II. Influence of salt, acid and flocculants." *Colloids and Surfaces*, 6(3), 239-251.

Lockhart, N. C., & Hart, G. H. (1988). "Electro-osmotic Dewatering of Fine Suspensions: The Efficacy of Current Interruptions." *Drying Technology*, 6(3), 415-423.

Luo, Q., Zhang, X., Wang, H., & Qian, Y. (2005). "Mobilization of Phenol and Dichlorophenol in Unsaturated Soils by Non-uniform Electrokinetics." *Chemosphere*, 59(9), 1289-1298.

MacDiarmid, A. G., & Ahmad, N. (1997). "Prevention of corrosion with polyaniline." U.S. Patent No. 5,645,890. Washington, DC: U.S. Patent and Trademark Office.

Malekzadeh, M., & Sivakugan, N. (2016). "Experimental Study on Intermittent Electroconsolidation of Singly and Doubly drained Dredged Sediments." *International Journal of Geotechnical Engineering*, 1-6.

Micic, S., Shang, J. Q., Lo, K. Y., Lee, Y. N., & Lee, S. W. (2001). Electrokinetic Strengthening of a Marine Sediment using Intermittent Current. *Canadian Geotechnical Journal*, 38(2), 287-302. doi:10.1139/t00-098.

Mitchell J. K., Soga K. (2005). Fundamentals of Soil Behavior (3rd edition).

Mitchell, G. F. (1976). The Irish Landscape. HarperCollins.

Mitchell, J. K. (1991). "Conduction Phenomena: from Theory to Geotechnical practice." *Géotechnique*, 41(3), 299-340.

Mitchell, J.K. (1993). Fundamentals of soil behavior. John Wiley & Sons, Inc., New York.

Mitchell, J.K., (1976). Fundamentals of soil behavior. John Wiley & Sons Inc.

Mohamedelhassan E, Shang JQ. (2002). "Feasibility assessment of electro-osmotic consolidation on marine sediment." *Proceedings of the ICE – Ground Improvement*, 6(4): 145-152.

Mohamedelhassan, E., and Shang, J. (2001). "Analysis of Electrokinetic Sedimentation of Dredged Welland River Sediment." *Journal of Hazardous Materials*, 85(1-2), 91-109. doi:10.1016/S03043894(01)00223-0.

Mosavat, N., Oh, E., and Chai, G. (2012). "A review of Electrokinetic Treatment Technique for Improving the Engineering Characteristics of Low Permeable Problematic Soils." *International journal of Geomaterial*, *2*(2), 266-272.

Napier, J. (1846). Philosophical Magazine. 29(10).

Nontananandh, S., Boonyong, S., Yoobanpot, T., and Chantawarangul, K. (2004). "Strength Development of Soft Marine Clay Stabilized with Cement and Fly Ash." *Natural Science - Kasetsart University*, 38(4), 539-552.

Ou, C. Y., Chien, S. C., & Chang, H. H. (2009). "Soil Improvement using Electroosmosis with the Injection of Chemical Solutions: Field tests." *Canadian Geotechnical Journal*, 46(6), 727-733. doi:10.1139/t09-012.

Ou, C. Y., Chien, S. C., & Wang, Y. G. (2009). "On the Enhancement of Electroosmotic Soil Improvement by the Injection of Saline Solutions." *Applied Clay Science*, 44(1), 130-136.

Ozkan S, G. R., Seals RK (1999). "Electrokinetic Stabilization of Saolinite by Injection of Al and PO43ions." *Ground Improvement*, 3, 135-144.

Perrin, J. (1904). The Journal of Chemical Physics. 2, 607.

Pugh, R. C., and Jones, C. J. F. P. (2004). U.S. Patent No. 6,736,568. Washington, DC: U.S. Patent and Trademark Office.

Quincke, G. (1861). "Ueber die fortfuhrung materieller theilchen durch strmende electricitat." *Annalen der Physik*, 113, 513-598.

Reddy, K. R., Urbanek, A., & Khodadoust, A. P. (2006). "Electroosmotic Dewatering of Dredged Sediments: Bench-scale Investigation." *Journal of Environmental Management*, 78(2), 200-208. doi:10.1016/j.jenvman.2005.04.018.

Reuss, F. F. (1809). "Memoires de la societe imperiale des naturalists de Moscow." 2, 327-337.

Richardson J.F., Zaki W.N. (1954). "Sedimentation and Fluidization." *Transactions of the Institute of Chemical Engineering*. 53, 32:35.

Rittirong, A., Shang, J., Mohamedelhassan, E., Ismail, M., and Randolph, M. (2008). "Effects of Electrode Configuration on Electrokinetic Stabilization for Caisson Anchors in Calcareous Sand." *Journal of Geotechnical and Geoenvironmental Engineering*. 134(3), 352-365. doi:10.1061/(ASCE)1090-0241(2008)134:3(352).

Sah, J. G., and Chen, J. Y. (1998). "Study of the Electrokinetic Process on Cd and Pb Spiked Soils." *Journal of Hazardous Materials*, 58(1–3), 301-315. doi:http://dx.doi.org/10.1016/S0304-3894(97)00140-4.

Sahib, A. A., and Vinod, P. (2010). "Evaluation of various electrode configuration in electroosmosis." *Proceeding of Indian geotechnical conference*, IGS Mumbai chapter and IT Bombay, 301-304.

Santos, J., Mattoso, L. and Motheo, A. (1998). Investigation of Corrosion Protection of Steel by Polyaniline Films. *Electrochimica Acta*, 43(3-4), 309-313.

Schaad, W., and Haefeli, R. (1947). "Elektrokinetische Erscheinungen und ihre Anwendungen in der Bodenmechanik." *Schweizerische Bauz&tung*, 65, 16-18.

Segall, B. A., and Bruell, C. J. (1992). "Electroosmotic Contaminant Removal Processes." *Journal of Environmental Engineering-Asce*, 118(1), 84-100. doi:10.1061/(asce)0733-9372(1992)118:1(84).

Shang, J. Q. (1997). "Electrokinetic Sedimentation: A Theoretical and Experimental Study." *Canadian Geotechnical Journal*, 34(2), 305-314.

Shang, J. Q. (1998). "Electokinetic Dewatering of Clay Slurries as Engineered Soil Covers." *Canadian Geotechnical Journal*, 34(1), 78-86. doi:10.1139/t96-083.

Shang, J. Q., & Dunlap, W. A. (1996). "Improvement of Soft Clays by High-voltage Electrokinetics." *Journal of geotechnical engineering*, *122*(4), 274-280.

Shang, J., & Dunlap, W. (1996). "Improvement of Soft Clays by high-voltage Electrokinetics." *Journal of Geotechnical Engineering*, 122(4), 274-280. doi:doi:10.1061/(ASCE)0733-9410(1996)122:4(274).

Sheets, K. R., Hendrickx, J.M.H. (1995). "Non-invasive soil water content measurement using electromagnetic induction." *Water Resources*, 31 (10), 2401–2409.

Skotheim, T. and Reynolds, J. (2007). Handbook of conducting polymers. Boca Raton, Fla.: CRC.

Smoluchowski, M. (1921). *Elektrische Endosmose und Strömungsströme*. Handbuch der Elektrizität und des Magnetismus, 2, 366-428.

Soderman, L. G., and Milligan, V. (1961). "Capacity of friction piles in varied clay increased by electroosmosis." *Proceedings of the 5th International Conference on Soil Mechanics and Foundation Engineering.* 2, 143-147.

Sprute, R. H., & Kelsh, D. J. (1980). "Dewatering fine-particle suspensions with direct current." In Proc. Int. Symp. Fine Particle Processes, 2, 1828-1844.

Sprute, R. H., and Kelsh, D. J. (1975). "Electrokinetic densification of hydraulic backfill: a field test." *Report of investigations (No. PB-247224; BM-RI-8075). Bureau of Mines, Spokane, Wash.(USA).* Spokane Mining Research Centre.

Stark, T. D., Choi H., and Schroeder P.R. (2005). "Settlement of Dredged and Contaminated Material Placement Areas. II: Primary Consolidation, Secondary Compression, and Desiccation of Dredged Fill Input Parameters." *Journal of waterway, port, coastal, and ocean engineering,* 2(52).

Supakij, N., Sanupong B. et al. (2004). "Strength Development of Soft Marine Clay Stabilized with Cement and Fly Ash." *Kasetsart Journal - Natural Science*, 38, 539-552.

Sverko, E. R. (1999). "Ground measuring techniques: electrode resistance to remote earth and soil resisitivity." *ERICO, Inc. Facility Electrical Protection, U.S.A.*

Veder, C. (1981). Landslides and their stabilization. by R. Hilbert, New York: Springer-Verlag.

Walker, I. (2008). "Using conductivity meters in agriculture." *Queensland government department of primary industries and fisheries available online.*

Yan, S., Singh, A. N., Fu, S., Liao, C., Wang, S., Li, Y., Hu, L. (2012). "A Soil Fauna Index for Assessing Soil Quality." *Soil Biology and Biochemistry.* 47(0), 158-165. doi:http://dx.doi.org/10.1016/j.soilbio.2011.11.014.

Yang, L., Nakhla, G., and Bassi, A. (2005). "Electro-kinetic Dewatering of Oily Sludges." *Journal of hazardous materials*, 125(1), 130-140.

Yeung, A. T., and Datla, S. (1995). "Fundamental Formulation of Electrokientic Extraction of Contaminant from Soil." *Canadian Geotechnical Journal*, 32(4), 569-583. doi:10.1139/t95-060.

Zhuang, Y., and Wang, Z. (2007). "Interface Electric Resistance of Electroosmotic Consolidation." *Journal of Geotechnical and Geo-Environmental Engineering*, 133(12), 1617–1621-1617–1621.